codex alimentarius commission

FOOD AND AGRICULTURE ORGANIZATION OF THE UNITED NATIONS

WORLD HEALTH ORGANIZATION

JOINT OFFICE: Via delle Terme di Caracalla 00100 ROME Tel.: 57971 Telex: 625852-625853 FAO I Cables: Foodagri Rome Facsimile: (6) 57973152-5782610

ALINORM 93/22

JOINT FAO/WHO FOOD STANDARDS PROGRAMME

CODEX ALIMENTARIUS COMMISSION

Twentieth Session Geneva 28 June - 7 July 1993

REPORT OF THE TWENTY-SECOND SESSION OF THE CODEX COMMITTEE ON FOOD LABELLING Ottawa, Canada, 26 - 30 April 1993

Note: This document incorporates Codex Circular Letter 1993/12-FL w/v0705

codex alimentarius commission

FOOD AND AGRICULTURE ORGANIZATION OF THE UNITED NATIONS

WORLD HEALTH ORGANIZATION

JOINT OFFICE: Via delle Terme di Caracalla 00100 ROME Tel.: 57971 Telex: 625852-625853 FAO I Cables: Foodagri Rome Facsimile: (6) 57973152-5782610

CX 4/15.2

CL 1993/12-FL May 1993

TO:

- Codex Contact Points

- Participants at the 22nd Session of the Codex Committee on Food Labelling

- Interested International Organizations

FROM:

- Secretary, Codex Alimentarius Commission, Joint FAO/WHO Food Standards

Programme, FAO, 00100 Rome, Italy

SUBJECT:

Distribution of the Report of the 22nd Session of the Committee on Food Labelling

(ALINORM 93/22)

A. MATTERS FOR ADOPTION BY THE 20th SESSION OF THE CODEX ALIMENTARIUS COMMISSION

Draft Guidelines at Step 8 of the Procedure

1. Draft Amendment to Section 3.3.4 of the Codex Guidelines on Nutrition Labelling (Draft Nutrient Reference Values for Labelling Purposes) (para. 23, Appendix II)

Governments wishing to propose amendments or comments on the above document should do so in writing in conformity with the Guide to the Consideration of Standards at Step 8 (see Procedural Manual of the Codex Alimentarius Commission) to the Secretary, Joint FAO/WHO Food Standards Programme, FAO, via delle Terme di Caracalla, 00100 Rome, Italy, before 20 June 1993.

Proposed Draft Guidelines at Step 5 of the Procedure

2. Proposed Draft Guidelines for the Production, Processing, Labelling and Marketing of Organically Produced Foods (para. 88, Appendix V).

Government wishing to submit comments on the implications which the above document may have for their economic interests should do so in writing in conformity with the Procedure for the Elaboration of Worldwide Standards at Step 5 to the Secretary, Joint FAO/WHO Food Standards Programme, FAO, via delle Terme di Caracalla, 00100 Rome, Italy, before 20 June 1993.

B. <u>DOCUMENTS TO BE ELABORATED FOR GOVERNMENT COMMENTS PRIOR TO</u> THE NEXT SESSION OF THE COMMITTEE

Proposed Draft Amendment at Step 3 of the Procedure

3. Proposed Draft Amendment to the Codex General Guidelines on Claims on the Use of the Term "Natural" (para. 95)

C. REQUEST FOR COMMENTS AND INFORMATION

Proposed Draft Guidelines at Step 3 of the Procedure

4. Propsed Draft Guidelines for the Use of Health and Nutrition Claims (paras. 48-49, Appendix III)

Governments are invited to submit specific comments on Sections 6. <u>Nutrient Function Claims</u> and 7. <u>Health Claims</u> and the relevant definitions thereof (points 2.1.3 and 2.2).

5. Proposed Draft Recommendations for the Labelling of Potential Allergens (Amendment to the General Standard for the Labelling of Prepackaged Foods) (para. 56, Appendix IV)

Governments are invited to submit comments on the proposals concerning labelling and the identification of potential allergens, and to provide scientific information on the occurrence of food allergies and their national approach to this question.

Other Matters

6. Implications of Biotechnology on International Foods Standards and Codes of Practice

Governments are invited to provide information on their national approach and policies with regard to the labelling of foods and food ingredients or additives produced through biotechnology.

Governments and international organizations wishing to submit comments and information on points 5. and 6. are invited to do so not later than 31 October 1993 to the Chairman of the Committee at the following address:

Mrs. K. Gourlie, Director Consumer Products Branch Consumer and Corporate Affairs 50 Victoria Street Hull, Quebec K1A OC9 Canada

with a copy to the Secretary, Joint FAO/WHO Food Standards, FAO, via delle Terme di Caracalla, 00100 Rome, Italy.

SUMMARY AND CONCLUSIONS

The summary and conclusions of the 22nd Session of the Codex Committee on Food Labelling are as follows:

Matters for consideration by the Commission:

The Committee:

- agreed to advance to Step 8 the Draft Proposal for the amendment of Section 3.3.4 of the Codex Guidelines on Nutrition Labelling (Draft Nutrient Reference Values for Labelling Purposes) (para. 23, Appendix II)
- agreed to advance to Step 5 the Proposed Draft Guidelines for the Production, Processing, Labelling and Marketing of Organically Produced Foods (para. 88, Appendix V)
- endorsed the labelling provisions of Proposed Draft (or Draft) Standards elaborated by CCTFFV, CCFFP, CCPL (para. 14) and recommended that CCNFSDU amend the labelling provisions of the Proposed Draft Standard for Formula Foods for Use in Very Low Energy Diets (at Step 5) (paras. 14-18)
- proposed that the Proposed Draft Guidelines for the Use of the Term "Natural" (at Step 3) become an amendment to the General Guidelines on Claims (para. 89)
- agreed to circulate at Step 3 the Recommendations for the Labelling of Potential Allergens in Foods (amendment to the General Standard for the Labelling of Prepackaged Foods) (para. 56, Appendix IV)

Other matters of interest to the Commission:

The Committee:

- agreed to return to Step 3 the Proposed Draft Guidelines for the Use of Health and Nutrition Claims for further comments and to have a new draft prepared by Canada for consideration by the next session of the Committee, in the light of the advice provided by CCNFSDU (paras. 48-49, Appendix III)
- agreed to return to Step 3 the Proposed Draft Guidelines for the Use of the Term "Natural" for redrafting by Canada as an amendment to the General Guidelines on Claims in the light of the discussions of the Committee (para. 95)

TABLE OF CONTENTS

	<u>Paragraph</u>
ntroduction	1
Opening of the Session	2-6
Adoption of the Agenda	7
Matters of interest arising from the Codex Alimentarius Commission and other Committees	8-11
Consideration of Labelling Provisions in Codex Standards	12-18
Proposed Draft Nutrient Reference Values for Food Labelling Purposes	. 19-23
Proposed Draft Guidelines for the Use of Health and Nutrition Claims in Food Product Labelling	24-49
Consideration of Labelling of Potential Allergens	50-56
Consideration of National Strategies regarding the Application of the List of Nutrients in the Guidelines on Nutrition Labelling	. 57-59
Revised Draft Guidelines for the Production, Processing, Labelling and Marketing of Organically/Biologically Produced Foods	60-88
Proposed Draft Guidelines for the Use of the Term "Natural"	89-98
Other Business and Future Work	99
Date and Place of the Next Session	100
LIST OF APPENDICES	<u>Page</u>
APPENDIX I - LIST OF PARTICIPANTS	17
- DRAFT NUTRIENT REFERENCE VALUES FOR FOOD LABELLING PURPOSES (Amendment to Section 3.3.4 of the Codex Guidelines on Nutrition Labelling)	36
PPENDIX III - PROPOSED DRAFT GUIDELINES FOR THE USE OF HEALTH AND NUTRITION CLAIMS	27
- CONSIDERATION OF POTENTIAL ALLERGENS IN FOODS (including Recommendations for the Labelling of Potential Allergens)	32
APPENDIX V - PROPOSED DRAFT GUIDELINES FOR THE PRODUCTION, PROCESSING, LABELLING AND MARKETING OF ORGANICALLY PRODUCED FOODS	56

INTRODUCTION

1. The Codex Committee on Food Labelling held its 22nd Session in Ottawa, Canada, from 26 to 30 April 1993 by courtesy of the Government of Canada. The Session was chaired by Mrs. Katharine Gourlie, Director, Consumer Products Branch, Consumer and Corporate Affairs Canada. The Session was attended by 124 participants, representing 29 member countries and 10 international organizations. A complete list of participants is given in Appendix I to this report.

OPENING OF THE SESSION (Agenda Item 1)

- 2. The Session was opened by the Honourable Pierre H. Vincent, Minister of Consumer and Corporate Affairs, who welcomed the delegates and observers to Ottawa on behalf of the Government of Canada.
- 3. Mr. Vincent noted that the 22nd Session of the Committee on Food Labelling fittingly coincided with Canada's National Consumer Week and that countries involved in producing or trading foodstuffs acknowledged that the role of consumers was now more dynamic and important than ever before. Consumers, industry and governments recognized the need for new national and international agreements in which it was essential to take into account consumers' concerns. Mr. Vincent stated that the Codex Alimentarius Commission had a leading role in serving and protecting the consumer, since it was responsible for establishing international standards for foods and ensuring fair practices in trade.
- 4. Mr. Vincent pointed out that the Canadian government was reviewing its consumer policies in response to a greater interdependence between consumers, industry and governments, and a constantly changing technological, economic and social environment. He noted that increased international trade gave consumers access to a greater variety of products, and that moreover they were increasingly aware of health concerns and looked for health and nutrition-related information on food labels such as the nutritional content.
- 5. Mr. Vincent also noted that the globalization of markets and the adoption of international trade agreements aimed at improving service to consumers, and that, when developing labelling requirements, governments wanted to ensure that label information was scientifically valid and easily understandable.
- 6. In conclusion, Mr. Vincent expressed his hope that the Committee's deliberations would result in greater international harmonization and agreement with regard to food labelling, and wished the Committee success in its work.

ADOPTION OF THE AGENDA (Agenda Item 2)

7. The Committee <u>agreed</u> to the adoption of the Provisional Agenda (CX/FL 93/1) as proposed. In order to facilitate its discussions concerning the consideration of Proposed Draft Guidelines for the Production, Processing, Labelling and Marketing of Organically and Biologically Produced Foods (Agenda Item 9), the Committee appointed an Ad Hoc Working Group to discuss this subject under the direction of Mrs. R. Lovisolo, FAO Consultant. The Committee also <u>agreed</u> to admit a representative of the Press (Food Chemical News) to the meeting, with the understanding that informal remarks presented by delegations would not be attributed to the Government concerned as an official position.

MATTERS OF INTEREST ARISING FROM CODEX ALIMENTARIUS COMMISSION AND OTHER CODEX COMMITTEES (Agenda Item 3)

8. The Committee had for its consideration documents CX/FL 93/2 and CX/FL 93/2-Add. 1 when discussing this agenda item, which summarized matters of interest arising from the Codex Alimentarius

Commission and other Codex Committees. The Secretariat also provided a verbal summary of the consideration of inspection and certification with regard to religious requirements, as discussed at the First Session of the Codex Committee on Food Import and Export Inspection and Certification Systems (paras. 48-50, ALINORM 93/30). As it was noted that most of the items in the working papers were for information only or were scheduled for discussion elsewhere, the Committee focused its discussions on the following matter.

Implications of Biotechnology on International Food Standards and Codes of Practice

- 9. As indicated in Conference Room Document 1 (CX/FL 93/2-Add. 1), the Committee was informed of the request of the 19th Session of the Codex Alimentarius Commission that CCFL should provide guidance on the possibilities to inform the consumer that a food had been produced through "modern" biotechnologies (paras. 88-92, ALINORM 91/40). The Committee also noted discussions held at the 25th Session of the Codex Committee on Food Additives and Contaminants concerning this subject, and especially that the competence of CCFL with respect to the labelling of food additives produced through biotechnology had been reasserted (paras. 81-93, ALINORM 93/12A).
- 10. In view of the complexity and importance of the issue of biotechnology as related to food labelling, the Committee welcomed the offer of the Delegation of the United States to prepare a discussion paper concerning this subject for circulation and government comments well before the next Session. It was also agreed that general comments and information on national policies concerning this issue would be requested by Circular Letter for consideration by the Delegation of the United States.
- 11. The Committee was further informed of the conclusions of the Committees on Food Hygiene, on Food Additives and Contaminants, on General Principles, on Food Import and Export Inspection and Certification Systems, on Nutrition and Foods for Special Dietary Uses, and on Methods of Analysis and Sampling. The Committee noted that the Table of Proposed Conditions for Claims for Nutrient Contents, agreed upon by CCNFSDU as part of the proposed Draft Guidelines on Nutrition and Health Claims for Food Product Labelling, would be considered under Agenda Item 6.

ENDORSEMENT OF LABELLING PROVISIONS IN CODEX STANDARDS (Agenda Item 4)

- 12. The Committee had for its consideration document CX/FL 93/3, containing labelling provisions submitted by various Codex Committees for endorsement in accordance with the Revised Codex General Standard for the Labelling of Prepackaged Foods (CODEX STAN 1-1985).
- 13. The Committee noted that the labelling provisions were being submitted in accordance with revised procedures concerning the Format for Codex Commodity Standards and Relations Between Commodity Committees and General Committees (pages 79 and 129-131, respectively, Codex Alimentarius Procedural Manual).
- 14. The Committee <u>endorsed</u> the labelling provisions of the following Codex Standards as submitted, with the understanding that those recommendations noted below would be considered by the Committee(s) concerned.

Codex Committee on Tropical Fresh Fruits and Vegetables 3rd Session, ALINORM 93/35

- Draft Standard for Pineapple (Step 8) (Appendix II)
- Draft Standard for Papaya (Step 8) (Appendix III)
- Draft Standard for Mango (Step 8) (Appendix IV)
- Proposed Draft Standard for Nopal (Step 5/8) (Appendix V)
- Proposed Draft Standard for Prickly Pear (Step 5/8) (Appendix VI)
- Proposed Draft Standard for Carambola (Step 5/8) (Appendix VII)

4th Session, ALINORM 93/35A

- Proposed Draft Standard for Litchi (Step 5/8) (Appendix II)
- Proposed Draft Standard for Baby Corn (Step 5/8) (Appendix III)
- Proposed Draft Standard for Banana (Step 5) (Appendix IV)
- Proposed Draft Standard for Avocado (Step 5) (Appendix V)

Codex Committee on Cereals, Pulses and Legumes

8th Session, ALINORM 93/29

- Proposed Draft Standard for Wheat (Step 5) (Appendix III)
- Proposed Draft Standard for Durum Wheat (Step 5) (Appendix IV)
- Proposed Draft Standard for Peanuts (Step 5) (Appendix V)
- Proposed Draft Standard for Oats (Step 5) (Appendix VI)

Codex Committee on Fish and Fishery Products

20th Session, ALINORM 93/18

- Proposed Draft Standard for Quick Frozen Squid (Step 5) (Appendix IV)
- Revised Codex Standard for Canned Shrimps or Prawns (Step 5) (Appendix V)
- Revised Codex Standard for Canned Salmon (Step 5) (Appendix VI)
- Revised Codex Standard for Canned Crab Meat (Step 5) (Appendix VII)
- Revised Codex Standard for Canned Sardines and Sardine-Type Products (Step 5) (Appendix VIII)
- Revised Codex Standard for Canned Tuna and Bonito (Step 5) (Appendix IX)
- Revised Codex Standard for Canned Finfish (Step 5) (Appendix X)
- Revised Codex Standard for Quick Frozen Shrimps or Prawns (Step 5) (Appendix XI)
- Revised Codex Standard for Quick Frozen Blocks of Fish Fillets, Minced Fish, Minced Fish Flesh and Mixtures of Minced Fish Flesh (Step 5) (Appendix XII)
- Revised Codex Standard for Quick Frozen Fish Sticks (Fish Fingers), Fish Portions and Fish Fillets Breaded or in Batter (Step 5) (Appendix XIII)
- Revised Codex Standard for Quick Frozen Lobsters (Step 5) (Appendix XIV)
- Revised Codex Standard for Quick Frozen Uneviscerated and Eviscerated Finfish (Step 5) (Appendix XV)
- 15. The Committee <u>agreed</u> to recommend that the CCFFP should simplify the labelling provisions of the above standards, as many of these requirements were already included in the General Labelling Standard. In other cases, it was also noted that some sections were missing (e.g., Sections 6.2 6.4, Codex Standard for Quick Frozen Shrimps and Prawns).
- 16. The Observer from the EEC informed the Committee about the observations they had made at the last session of CCFFP; certain provisions in the above Draft Standards differed from the EEC regulations in this area, especially the indication of net weight of glazed products, the Scope of the Standard for Sardine and Sardine-type product and the association of the terms tuna and bonito on the label in the Revised Codex Standard for Tuna and Bonito and that these observations had been made at the last session of CCFFP. The Delegation of France reserved their position on the endorsement of the labelling provisions of several of the above Quick Frozen Products as the Net Content provisions for Glazing differed from EEC requirements. France also reserved their position on the use of the dual product name "Bonito-Tuna" in the Standard for Canned Tuna and Bonito as well as the use of the product name "Sardines" for Pilchard Sardines. It was agreed that these comments would be communicated to the Committee on Fish and Fishery Products.

17. With regard to the suggestions concerning the addition of species to the standards, it was noted that these comments could be directed to the CCFFP at Step 6.

Codex Committee on Nutrition and Foods for Special Dietary Uses

18th Session, ALINORM 93/26

- Proposed Draft Standard for Formula Foods for Use in Very Low Energy Diets (Step 5) (Appendix II)
- 18. The Committee <u>agreed</u> to recommend that the CCNFSDU should remove sections 9.2 and 9.4 of the proposed draft labelling section, as these provisions were already covered under the General Standard for the Labelling of and Claims for Prepackaged Foods for Special Dietary Uses (CODEX STAN 146-1985). The Committee also recommended that when referring to Sections 9.1 and 9.6 in section 9.7, the applicable requirements should be explicitly indicated, as it was not clear which labelling provisions applied to the label of the package and/or the sachet, and which statements might appear on an accompanying leaflet.

PROPOSED DRAFT NUTRIENT REFERENCE VALUES FOR FOOD LABELLING PURPOSES (Agenda Item 5)

- 19. The Committee had for its consideration Appendix V of ALINORM 91/22, presenting a draft amendment to Section 3.3.4 of the Codex Guidelines on Nutrition Labelling (i.e. Draft Nutrient Reference Values) as adopted at Step 5 by the 19th Session of the Commission. Government comments at Step 6 in reply to CL 1992-5/FL were contained in document CX/FL 93/4 (Austria, Denmark, New Zealand, United States) and CX/FL 93/4-Add.1 (Malaysia). A document containing extracts from a report of the Scientific Committee for Food of the EEC on "Nutrient and Energy Intakes for the European Community" was distributed as an unnumbered Room Document.
- 20. The Committee recalled that Section 3.3.4 of the Codex Guidelines on Nutrition Labelling had been amended by incorporating the Nutrient Reference Values (NRVs) recommended by the Joint FAO/WHO Expert Consultation on Recommended Allowances of Nutrients for Food Labelling Purposes (1988, Helsinki). The Committee noted that it was not competent to set reference values for recommended nutritional intake and that expert advice should be sought on such aspects, and that the values under consideration were intended for labelling purposes only.
- 21. Some delegations pointed out that as a number of source materials existed, with different values for women and children in certain cases, appropriate guidelines or principles would be needed to ensure that the establishment of NRVs was based on appropriate scientific data. Moreover, new evaluations were currently being carried out at the national and international levels and significant evolutions had taken place in recent years in this particular field. In this perspective, some delegations suggested that the Nutrient Reference Values should not be adopted at the present time but that further guidance should be sought from the Committee on Nutrition and Foods for Special Dietary Uses, especially as regarded principles. Other delegations stressed that these values had been proposed as a result of the Helsinki expert consultation and were supported by CCNFSDU. It was well understood that the definition and review of such values was an ongoing process, subject to revision in the light of new scientific data, as the footnote to the list of NRVs presented in Appendix V of ALINORM 91/22 clearly indicated. The Committee was also reminded that the last Session of CCNFSDU had requested government comments on this matter, for consideration at its next Session (September 1994).
- 22. While recognizing the need for general principles to guide the choice and amendment of reference values, the Committee felt that the adoption of the revised list, which represented an updating of the previous list, should not be delayed until such time as these principles were elaborated. The Committee therefore agreed to advance to Step 8 the draft reference values and to request the guidance of CCNFSDU

on guidelines which should be followed to establish and revise such reference values. The Committee was of the opinion that an indication of the purpose for which these guidelines were needed should be given to CCNFSDU. After an extensive exchange of views on this issue, the Committee agreed that the following elements should be taken into consideration by CCNFSDU when elaborating "Principles for the Consideration of NRVs for Labelling Purposes":

- 1. While establishing Nutrient Reference Values (NRVs), the Principles for Nutrition Labelling as contained in the Codex Guidelines on Nutrition Labelling should be followed.
- 2. The primary purpose of nutrient reference values is to give meaningful and not misleading information to the consumer.
- 3. The composition of the list should be carefully considered and justified in terms of the consumers' needs.
- 4. A list of NRVs for foods targeted to the general population should be established. The need to establish NRVs for specific groups such as infants and children should also be considered.
- 5. NRVs should be based as far as possible on nutrient intakes recommended by FAO and/or WHO. The source of values should be indicated and justified.

Status of the Draft Nutrient Reference Values for Food Labelling Purposes (Proposal for the amendment of Section 3.3.4 of the Codex Guidelines on Nutrition Labelling)

23. The Committee <u>agreed</u> to advance the aforesaid Draft NRVs, as contained in Appendix II of the present report, to Step 8 of the Procedure for adoption by the Commission.

PROPOSED DRAFT GUIDELINES FOR THE USE OF HEALTH AND NUTRITION CLAIMS IN FOOD PRODUCT LABELLING (Agenda Item 6)

- 24. The Delegation of Canada introduced working paper CX/FL 93/6, containing the Proposed Draft Guidelines, which took into account the proposals of the previous session of the Committee. Government comments were presented in document CX/FL 93/6-Add. 1 (Denmark, Poland), Add.2 (Malaysia, Sweden, International Special Dietary Foods Industries), Add. 3 (International Dairy Federation), Add.4 (Joint Nordic comments of Finland, Iceland, Norway and Sweden), Add.5 (Brazil), Add.6 (International Organization of Consumers Unions).
- 25. The Committee <u>agreed</u> to amend the title of the Guidelines so as to harmonize it with the "General Guidelines on Claims" by removing "in food product labelling", and pointed out that the terms of reference of the Committe explicitly referred to Food Labelling.

Section 1. SCOPE

26. Some delegations expressed the view that the Guidelines should not include foods for special dietary uses, as specific provisions applied in their case. Other delegations pointed out that claims of a general nature relating to nutrient content could still be made for these foods. The Committee agreed to indicate that the Guidelines applied to all foods "without prejudice to specific provisions under Codex standards or guidelines relating to Foods for Special Dietary Uses and Foods for Special Medical Purposes".

Section 2. DEFINITIONS

27. The Committee decided to use the wording of the General Guidelines on Nutrition Labelling for the definition of Nutrition Claim in point 2.1., and to delete the reference to "standardized terminology"

- in point 2.1.1. It was further agreed that energy value should be specifically mentioned in the definition of comparative claims (point 2.1.2).
- 28. The Committee <u>agreed</u> to amend the definition of Nutrient Function Claim so as to indicate that "it sets out in general terms the nutritional consequences for good health of the intake of a particular nutrient".
- 29. The Committee had an exchange of views on the conditions under which health claims relating to food might be used, and discussed a proposal to add in 2.2 <u>Health claims</u> "that this relationship be expressed only within the context of the total diet".
- 30. With respect to the definitions of Nutrient Function Claim and Health claim, some delegations expressed the view that the distinction between these types of claims was very difficult to establish and that they should be considered jointly. Other delegations were of the opinion that distinct definitions should be retained.
- 31. The observer from the EEC informed the Committee that claims relating to the cure of diseases were prohibited in the Community except in the framework of the specific legislation on mineral waters and foods for special dietary uses. Some delegations indicated that health claims establishing relationships between a nutrient and the prevention of a disease could be allowed within the context of a total dietary pattern. Other delegations did not accept any reference to diseases.

Section 4. NUTRIENT CONTENTS CLAIMS

- 32. Some delegations questioned the intention expressed in point 4.1 as to claims which did not fall within the scope of the definition and the Table, judging that it was too restrictive to limit nutrient content claims to the descriptors proposed. They pointed out that reference might be made to a percentage or quantity of a nutrient, without qualifying it as «low» or «high» for example, or without any other claim. The Committee therefore agreed to amend the definition as follows, to make it clear that the values indicated in the Table applied to the specific claims presented therein:
 - «When a nutrient content claim that is listed in the Table to these Guidelines or a synonymous claim is made, the conditions specified in the Table for that claim should apply.»
- 33. The Committee had an exchange of views on point 4.2 and the Delegation of Norway, on behalf of the Nordic countries, proposed that claims should be permitted only when the nutrient was naturally present in the food, in order not to mislead the consumer. The Committee did not accept this proposal, as such a provision appeared too restrictive.

Section 5. COMPARATIVE CLAIMS

- 34. On the suggestion of the observer from IOCU, the Committee <u>agreed</u> to indicate that claims applied to the end-products as sold and not to their ingredients.
- 35. In point 5.1, the Committee <u>agreed</u> that «foods which replace each other in the diet» was not explicit enough and that reference should be made to «different versions of the same food or similar foods».
- 36. In point 5.2.1, the Committee considered the requirements which should apply when the claim referred to an absolute amount, as opposed to a percentage or fraction. Some delegations stressed that the consumer should not be mislead and questioned whether the total value of the nutrient present in the food should be indicated so as to provide an adequate basis for comparison or if any absolute amount should refer to 100g of the food. After an exchange of views on this question, the Committee <u>agreed</u> to refer to "the amount of difference, related to the same quantity, expressed as a percentage, a fraction or an absolute

amount», to delete the second sentence and to replace it with a general mention covering all possible cases, to the effect that «Full details of the comparison should be given».

- 37. With respect to point 5.3, some delegations were of the opinion that the relative difference should be increased to allow a comparative claim. The Committee noted that the differences should be nutritionally significant and that this aspect was addressed in the last part of 5.3, referring to a minimum absolute difference.
- 38. Some delegations pointed out that the percentages used for the purposes of comparison should be different for macronutrients and micronutrients. It was also suggested that a numerical indication should be permitted for a different amount of reduction provided no qualifying claim was made. Some delegations indicated that provisions should be made for the use of the term "light", when used as a claim for reduction. The Committee agreed to take these proposals into account in point 5.3 and to add a new point 5.4 in square brackets:

[«the use of «reduced» (e.g. light) or «increased» should be restricted to changes of at least 25% of energy or macronutrients or 10% of the NRVs for micronutrients. This should not preclude factual numerical statements about smaller changes.»]

The Observer from IOCU noted that claims relating to "goodness" such as "good for you", "full of goodness" were common health claims and needed to be included in the Guidelines. The Committee agreed to consider this question at the next session.

Section 6. NUTRIENT FUNCTION CLAIMS

- 39. The Committee <u>agreed</u> to add a new point 6.1.3 to the effect that "the food for which the claim is made should be a significant source of the nutrient in the diet", so as to avoid misleading the consumer.
- 40. Some delegations expressed the view that such claims should not be allowed. It was also pointed out that point 6.1 was similar to point 7.1 relating to health claims and that there was no clear distinction between these definitions. The Committee noted that clarification was needed as to the amount of a nutrient necessary to justify such a claim and the definition of the function of nutrients, and agreed to refer this question to CCNFSDU for guidance. The Delegation of the United Kingdom indicated that it appeared necessary to merge Sections 6. Nutrient Function Claims and 7. Health Claims and the Committee agreed to request specific government comments on this issue, and noted that no consensus could be reached at this stage.

Section 7. HEALTH CLAIMS

- 41. The Chairman pointed out that the provisions of point 7.2 were already included in the General Guidelines on Claims (Section 3.4), and the Committee <u>agreed</u> to delete it.
- 42. Some delegations were of the opinion that health related claims could be allowed in relation to a total dietary pattern and if they could be substantiated by scientific evidence. Other delegations and the observer of the EEC expressed the view that no claim establishing a relation between a food or nutrient and a disease or health-related condition should be allowed. The Committee had an extensive exchange of views on this question but could not reach a consensus on the claims which should be allowed. The Committee consequently agreed to ask the advice of CCNSDFU on the health effects of nutrients as specifically related to claims, (Points 7.1 and 7.3). It was further agreed to establish a drafting group including Australia, Canada, Denmark, France, Japan, New Zealand, Norway, Sweden, Switzerland, United Kingdom, United States, CIAA, IDF, IFGMA, IOCU, under the direction of the Delegation of Canada, to consider the provisions of Sections 6. and 7. and the related definitions (points 2.1.3 and 2.2) and prepare a revised proposal for consideration by the next Session.

Section 8. CLAIMS RELATED TO DIETARY GUIDELINES OR HEALTHY DIETS

43. On the suggestion of the Delegation of Germany, the Committee <u>agreed</u> to refer to "a national authority" and delete "having jurisdiction", as dietary guidelines might be established by scientific bodies with no regulatory power. On the suggestion of the observer of the IOCU, the Committee <u>agreed</u> to amend the wording of 8.4 to make it more explicit.

TABLE

- 44. The Secretariat indicated that CCFL was not mandated to amend the values in the Table, as the responsibility of establishing the values of nutrients relating to claims rested with CCNFSDU, as indicated by the 36th Session of the Executive Committee, and that the comments of the Committee would be communicated to CCNFSDU.
- 45. Several delegations made comments on the figures proposed to describe claims, especially on the following aspects: most references in the Table were expressed in relation to a fixed quantity of food, whereas the reference to serving was often more useful to the consumer; all nutrient values were not expressed with the same criteria and more consistency should be sought in this matter; modifications were suggested to the figures for "free", "source" or "high" for certain nutrients. Some delegations were of the opinion that "cholesterol free" should be retained, as Codex guidance was needed for this type of claim when used. The Delegation of the United Kingdom pointed out that in view of the changes proposed to point 4.1, the Table would no longer be exclusive and therefore where no criteria were indicated, the industry would be free to choose their own. The Delegation of Switzerland suggested to consider specifically the following values: 0.5g for "fat free"; 1g for "sugars free"; for vitamins and minerals, 15% of the NRV for "source" and 50% for "high". The Delegation of Norway suggested that the conditions for claims should refer to the energy unit (e.g. per 100kcal) instead of a fixed quantity or a serving. In reply to these concerns, the Secretariat recalled the last CCNFSDU Session had come to its conclusions after detailed consideration of these issues; in particular, it had been decided to delete the reference to "cholesterol free" as of no nutritional significance, and that the level of cholesterol in a food should be considered in relation to the energy derived from saturated fats. The Committee was also informed that the proposed values reflected the compromise reached between different approaches, especially as to the description of "free", and that many levels were indicated in square brackets and might be revised in the light of further comments. The Committee agreed that CCNFSDU should be informed of these comments, so as to reconsider the nutrient contents proposed if necessary.
- 46. The observer of the IDF expressed concern regarding the requirements proposed for "low", pointing out that provisions relating to fat could not be met for dairy products where the fat content was originally very high, as a reduction of 50% was already very significant and consumers should be informed of it. It was also pointed out that Codex standards or Draft standards referring to "low fat" already existed. The Delegations of Denmark and Norway as well as the Observer from IOCU, were of the opinion that the same requirements for "low" should apply to all foods.
- 47. The Committee noted that a Proposed Draft Standard for Fat Spreads was scheduled for consideration by the next Session of the Committee on Fats and Oils (September 1993), and that questions specifically related to dairy products would be considered by the Committee on Milk and Milk Products.
- 48. The Committee <u>decided</u> that as agreement could not be reached regarding the nutrient function claims and health claims, the Proposed Draft Guidelines should remain at Step 3 and be circulated for further comments. It was agreed to append to the report the text as amended by the Committee, in the sections where consensus existed. The Drafting Group, under the direction of the Delegation of Canada, would consider Sections 6. and 7. and the related definitions (point 2.1.2 and 2.3) in order to prepare a revised proposal for consideration by the next session. The Committee further <u>agreed</u> to seek the advice and guidance of CCNFSDU on these sections, and to inform CCNFSDU of the comments made with respect to the Table of nutrients.

Status of the Proposed Draft Guidelines for Use of Nutrition and Health Claims

49. The Committee <u>agreed</u> to return the Proposed Draft Guidelines, as contained in Appendix III to Step 3 for further comments.

CONSIDERATION OF LABELLING OF POTENTIAL ALLERGENS (Agenda Item 7)

- 50. The Committee had before it document CX/FL 93/5, which was prepared by Norway on the above subject in co-operation with Finland, Iceland and Sweden. Comments submitted by Austria were summarized in an unnumbered Conference Room Document.
- 51. The Committee recalled its previous discussions concerning this subject, whereby it was decided to examine the labelling of potential allergens which were included as components of composite ingredients in foods, and thus were not included in the ingredients list. The Committee noted that this issue would require the examination of Section 4.2.1.3 of the General Labelling Standard which addressed the labelling of composite ingredients, especially as related to the 25% rule (paras. 146-147, ALINORM 91/22).
- 52. The Delegation of Norway introduced the paper, which included a detailed summary of information concerning hypersensitivity (allergy and intolerance) problems and improvements associated with labelling in relation to hypersensitivity, issues related to the carry-over of food additives into foods and recommendations concerning amendments to the General Labelling Standard. The Delegation pointed out that labelling with regard to potential allergens should not give a false sense of security to affected consumers; however, although practical solutions might be difficult to find in this respect, the problems of hypersensitive consumers could not be solved without adequate labelling, which in itself provided a essential basis for other or complementary measures.
- 53. Several delegations noted that the matter of food hypersensitivity was both a public health and labelling issue, and that hypersensitive consumers relied on labelling to avoid certain foods. The Delegation of the United States pointed out in particular that the 25% rule should be revised. Certain delegations stated that the principle of complete and accurate ingredient declarations was generally accepted by governments and industry alike, but that additional industry co-operation was required, especially where second and third generation ingredients of foods were concerned. It was also noted that the recent Joint FAO/WHO International Conference on Nutrition recognized problems associated with food allergies, especially as related to infants and children.
- 54. A number of delegations suggested that actions required should be relative to the amount of risk, and that attempts should be made to define the ingredients and additives involved, as a balance was required to provide adequate, as opposed to burdensome, labelling information. The difficulties of establishing guidelines on a worldwide basis were also noted, as food allergies were often associated with specific sub-groups or regions. The relative costs associated with labelling and the categorization of potential allergens were also highlighted, as was the importance of justifying any changes to composite labelling requirements with appropriate criteria resting on a scientific basis. It was also noted that care should be taken not to create a sense of false security by action on labelling.
- 55. The Observer from the Association of European Celiac Societies (AOECS) noted the importance of labelling information for gluten sensitive individuals, as this group required in addition to the special gluten-free diet products, totally gluten-free foods for normal consumption. The Observer stated that detailed information was necessary because gluten was often added to foodstuffs where it normally was not expected (e.g., ham), or was part of a composite ingredient which was not required to be sub-listed in the ingredients list. The identification of ingredients such as starch should also be more explicit (e.g. wheat starch).

56. The Committee, while noting the importance of continuing discussions concerning this subject, agreed to append the recommendations for the amendment of the General Standard for the Labelling of Prepackaged Foods and the attached background document to the present report, as contained in Appendix IV, for circulation and government comments at Step 3.

CONSIDERATION OF NATIONAL STRATEGIES REGARDING THE APPLICATION OF THE LIST OF NUTRIENTS IN THE CODEX GUIDELINES ON NUTRITION LABELLING (Agenda Item 8)

- 57. The Committee had for its consideration document CX/FL 93/7 when discussing this agenda item, which contained government comments submitted by the United States in response to CL 1991/11-FL.
- 58. The Committee recalled that at its previous session it was agreed to request government comments on the type of nutrients they considered to be relevant for maintaining good nutritional status, as required by national legislation.
- 59. The Observer of the EEC informed the Committee that in application of Directive 90/496/EEC (24 September 1990), when nutrition labelling was provided, the following information would be required:
 - As of 1 October 1993: energy value, protein, carbohydrate and fat;
 - As of 1 October 1995: sugars, saturated fat, fiber and sodium (i.e., in addition to the above categories) whenever voluntary nutrition labelling existed or a nutrition claim was made for one of these nutrients.

REVISED PROPOSED DRAFT GUIDELINES FOR THE PRODUCTION, PROCESSING, LABELLING AND MARKETING OF ORGANICALLY/BIOLOGICALLY PRODUCED FOODS (Agenda Item 9)

- 60. The Committee had for its consideration document CX/FL 93/8 presenting the comments (Cuba, Denmark, Germany, Japan, New Zealand, Spain, United States, AMFEP, EEC, IFOAM) received in reply to CL 1991/23-FL, containing the first draft of the Guidelines (ALINORM 91/37), document CX/FL 93/8-Add. 1, which was a revised version of the Proposed Draft Guidelines for the Production, Processing, Labelling and Marketing of Organically/ Biologically Produced Foods, amended on the basis of the comments. The Committee also considered comments on the current draft as contained in CRD 2 (IFOAM) and CRD 3 (Japan).
- 61. The Committee was informed of discussions held at the 19th Session of the Commission concerning this issue, whereby it was agreed that the CCFL would be responsible for the future development of the guidelines presented (ALINORM 91/37) based on comments received from governments and Codex Regional Coordinating Committees (para. 103, ALINORM 91/40). The Committee also noted discussions held at the Coordinating Committees for North America and the South-West Pacific (paras. 87-90, ALINORM 93/32) and for Europe (paras. 80-84 and 88-89, ALINORM 93/19) concerning this subject.
- 62. The current Guidelines (CL/FL 93/8-Add.1) were presented by Ms. R. Lovisolo, FAO Consultant, who gave a brief summary of the background and elements suggested for organic foods. In this regard, several delegations noted that the work of the CCFL should be restricted to labelling, as other elements of the guidelines more appropriately rested with other Committees, such as the Codex Committee on Food Import and Export Inspection and Certification and the Codex Committee on Food Additives and Contaminants. It was also suggested that each country should decide whether it was appropriate to establish mandatory requirements for organic foods, while other delegations noted that the guidelines, as applied to international trade, were of an advisory nature.

- 63. The Committee agreed that its primary responsibilities were related to labelling and similar areas, as it was recognized that certain aspects of this work would be addressed more adequately by other Codex Committees.
- 64. The Committee also agreed to discuss the recommendations contained in the report of the Working Group on Organics which was chaired by the FAO Consultant and attended by Australia, Canada, France, Germany, Japan, Lithuania, Netherlands, Norway, Sweden, Switzerland, United States, EEC, IFOAM and IOCU. As a result of these discussions, the Committee agreed to the following changes to the Revised Proposed Draft Guidelines (CX/FL 93/8-Add.1):

SECTION 1. SCOPE

65. Paragraph 1.1 (b) should read:

processed product for human consumption derived mainly from (a) above.

as there are organically-produced products destined for purposes other than human consumption.

66. The term "bio-dynamic" or reference to other specific terms such as "biological" should be included only in a list of reserved terms and the square brackets be deleted as this would prevent deceptive labelling. The title of the Guidelines would therefore refer only to "organically produced foods". Paragraph 1.2 (b) should also be deleted.

SECTION 2. DESCRIPTION AND DEFINITIONS

67. As the term "agricultural origin" (paras. 3.3 (b); 3.4; 3.4 (a) and (b); and 3.6 (a) and (b)) could be applied differently between countries, (eg. some countries exclude fish products from the term) the definition for "agricultural product" was modified as follows:

"agricultural product" means any product or commodity, raw or processed, that is marketed for human consumption.

It was suggested that a definition for the term "veterinary drug" should be provided.

68. It was agreed that 2.2 (c) be altered to read:

"marketing" means holding for sale or displaying for sale

in order to clarify the definition. The definition for the term "bio-dynamic" should be removed as this term should not be isolated in this regard.

SECTION 3. LABELLING

69. As a means of protecting consumers against deceptive practices, a restriction on the blending of an ingredient derived from an organic and non-organic source within a single product should be included in the guidelines. The following statement was included in paragraph 3.3, between the present sub-points (c) and (d):

The same ingredients shall not be derived from both organic and non-organic origin.

- 70. The substance of paragraphs 3.4 (a) and (b) were amalgamated to read:
 - are of agricultural origin and are not produced in the country or in sufficient quantity in the country in accordance with the requirements of Section 4 of these guidelines.

- 71. As promotional limitations could be imposed on the industry by paragraph 3.6 in that product containing up to 50% organic ingredients should be restricted to using organic product indications solely on the ingredients list, specific country comments should be sought on the feasibility of allowing other descriptive statements on products containing between 50% and 95% of organically-derived ingredients.
- 72. Paragraph 3.7 relating to conversion labelling was deleted as there would be limited advantage in utilizing such labelling provisions that could add new potential labels likely to be confusing to consumers.

SECTION 4. RULES OF PRODUCTION

73. Paragraph 4.2 covered an aspect of the specific principles of organic production and was moved to Annex 1.

SECTION 5. REQUIREMENTS FOR INCLUSION OF SUBSTANCES IN ANNEX 2

- 74. Point 4 of Annex 2 was moved to Section 5, paragraph 5.1 (d), third dash, as this should be a criterion for determining substances included on national lists, as follows:
 - they are preferably nature identical and it is impossible to produce or preserve such food products without having recourse to such ingredients.
- 75. The guideline should include criteria for the inclusion of substances in national lists which was covered by paragraph 5.1. This was, therefore, amended as follows:

Countries should develop a list of substances which satisfy the requirements of this guideline. The following criteria should be used for the purposes of amending these lists of substances not authorized for the purposes indicated in Section 4.

As a consequence of the amendment to paragraph 5.1, paragraph 5.2 would set out the criteria for countries to make application to Codex for additions or amendments to the Codex list. Paragraph 5.2 should then to read:

Member countries should provide the following for any substance proposed for inclusion in Annex 2.

SECTION 6. INSPECTION AND CERTIFICATION SYSTEMS

76. The use of the term "country" in paragraphs 6.2, 6.4 and 6.5 should be changed to read: "competent authority" in paragraph 6.2 and "country authority" in paragraph 6.4 and 6.5. As the designated authority responsible for the approval and supervision of private bodies should be able to delegate this function to an agent an additional provision should be included:

The authority may delegate this function to its agent.

SECTION 7. INDICATIONS THAT PRODUCTS ARE COVERED BY AN INSPECTION SYSTEM

77. This section of the revised draft guidelines which related to provision for the use of a logo was considered not appropriate in the context of the Codex guidelines and was deleted.

SECTION 8. IMPORTS (becomes Section 7.)

78. The principle of equivalency within the guidelines may need further amplification, and should be considered in the light of further comments. It was considered appropriate to include for further discussion

the IFOAM proposal that countries might rely on non-governmental organizations for the determination of bilateral equivalency decisions.

ANNEX 1: PRINCIPLES OF ORGANIC PRODUCTION

Plants and Plant Products

79. It was recommended that the square brackets should be retained in paragraphs 1 and 2 so as to provide all countries with an opportunity for further comments.

Livestock Production

80. Paragraph 7 would be too restrictive in terms of requiring that livestock be maintained as part of an organic farm unit and was redrafted as follows:

Where livestock are maintained, they should be an integral part of the organic farm unit and should be raised and held according to these guidelines.

- 81. In order to clarify conversion aspects of livestock production paragraph 9 was deleted and the following included:
 - 9. Animal products must not be sold as organic unless the animal has been raised according to these guidelines for a period of at least one year.
 - 10. Up to 10% of adult animals of a herd or flock may be brought-in annually from nonorganic sources for expansion or replacement purposes.
 - 11. All brought-in animals from non-organic sources must be produced according to these guidelines for a period of a minimum of one year before their products may be sold under an organic label. Exceptions may be allowed for:
 - (a) calves up to 14 (or 7?) days which have received colostrum and do not come from livestock markets;
 - (b) dairy animals provided that milk is kept separate for a period of 12 (or 4?) weeks;
 - (c) day old poultry; and
 - (d) laying hens, provided that eggs are kept separate for a period of 30 days.

Processing, Storage and Transport

82. A section on processing, storage and transport was developed and included in square brackets in the amended draft guidelines. Specific provisions for transport would not be necessary as national governments have extensive controls in place in this regard.

ANNEX 2. PERMITTED SUBSTANCES FOR THE PRODUCTION OF ORGANIC FOODS

83. Paragraph 4 was moved to 5.1(d) third dash point. The feasibility of an approach based on generic categories of substances as an alternative to the development of lists was considered. As the structure of the proposed lists had already achieved a substantial degree of consensus, it was recommended that further comments should be sought on this matter.

ANNEX 3. MINIMUM INSPECTION REQUIREMENTS AND PRECAUTIONARY MEASURES UNDER THE INSPECTION SYSTEM

84. Paragraph 8 should be amended as follows to prevent confusion:

" ... should be transported in a manner which would prevent"

Status of the Proposed Draft Guidelines on the Production, Processing, Labelling and Marketing of Organically/Biologically Produced Foods

85. The Committee agreed that the Proposed Draft Guidelines, as contained in Appendix V, would be advanced to Step 5 for adoption by the 20th Session of the Commission. This decision was made with the understanding that other relevant Codex committees would be informed.

<u>CONSIDERATION OF PROPOSED DRAFT GUIDELINES FOR THE USE OF THE TERM "NATURAL"</u> (Agenda Item 10)

- 86. The Committee had before it the Proposed Draft Guidelines for the Use of the Term "Natural" (CL 1992/28-FL) as prepared by Canada, as well as comments submitted at Step 3 on this proposal, presented in documents CX/FL 93/9 (Denmark, Spain), CX/FL 93/9-Add. 1 (Australia), Add.2 (Switzerland). Add.3 (USA), Add.4 (GISEM-UNISEM European Union of Natural Mineral Water Sources of the Common Market), Add.5 (Sweden), Add.6 (IDF)
- 87. The Committee recalled that the proposed draft guidelines were first considered at its 21st Session (paras. 87-106, ALINORM 91/22), after which Canada prepared revised guidelines based on written comments and discussions at the meeting. While introducing the revised guidelines, Canada noted that its provisions were significantly more restrictive than the first draft.
- 88. Several delegations stated that they did not support the development of guidelines for single words such as "natural", as it was felt that the use of such terms were already addressed by provisions in the Codex General Guidelines for Claims (i.e, Section 5.1 (iii)). Differences in the meaning of "natural" in various languages were also noted as a problem in the development of the guidelines, as were the difficulties in establishing exact meanings. The Committee, however, favoured preparation of guidelines in view of the widespread use of this term and the potential to mislead the consumer.
- 89. The Committee considered several options on how to proceed and <u>decided</u> to amend the General Guidelines for Claims with a new section on the use of the term "natural", as opposed to the continued development of separate guidelines, with the understanding that the Scope of the document would remain inchanged. It was further noted that Section 5.1 (iii) (Conditional Claims) of the General Guidelines, referring to "natural", would have to be amended in consequence. The Committee also <u>agreed</u> to discontinue the consideration of a definition for "traditional processes" and as a consequence, removed all references to this term from the text.
- 90. The Committee <u>accepted</u> the offer of the Delegation of Canada to elaborate amendments to the Codex General Guidelines on Claims relating to the use of the term "natural". In this regard, it was <u>agreed</u> that written comments received on CL 1992/28-FL as well as the following discussions would be taken into consideration when preparing these amendments.

Section 2. Criteria for the Use of "Natural"

91. In point 2.1.2, the Committee noted that irradiated foods would not qualify for natural labelling. The Delegation of Germany pointed out that provisions concerning pesticides and veterinary drugs should be included. The Committee agreed to remove the square brackets around the terms [food additives], [vitamins], [minerals], [colors] and [flavors] in view of their prohibition for use in "natural foods".

Section 3. Additional Labelling Requirements

92. The Committee <u>agreed</u> that the requirement set out in point 3.2 should not be made in view of the difficulties in including such a burdensome statement on product labels and the possibility to mislead consumers.

Minimal Processes

- 93. Several delegations took the position that processes such as pasteurization, roasting, cooking, blanching and baking should be considered as "minimal" as they were necessary in many cases to make the food fit for human consumption. Other delegations felt that such processes should be limited to those that involved minimal manipulation of the food and that cooking and pasteurization should be excluded. The Committee agreed to add the following words to the end of the definition of minimal processes: "... and include those processes necessary to make the food fit for human consumption." The Committee had an exchange of views on the examples to be included; it was also suggested that they should not be included in the definition, as a consensus could not be reached. The observer from I.D.F. stressed the need for the Committee to address the definition of permitted processes which would otherwise be left to individual countries. The Delegations of Norway, Sweden and Switzerland disagreed with this decision as they regarded the definition of minimal processes as too wide.
- 94. It was pointed out that specific provisions existed for mineral waters and that a statement on the exclusion of specific Codex standards should be included.
- 95. The Committee <u>agreed</u> that the proposed draft amendments to the General Guidelines on Claims on the use of the term "natural" would be prepared by Canada, on the basis of the comments received and the above discussion and circulated for further comments at Step 3 prior to the next session.

OTHER BUSINESS AND FUTURE WORK (Agenda Item 11)

- 96. The Committee noted that the following matters would be considered at its 23rd Session:
 - Consideration of Labelling Provisions in Codex Standards;
 - Proposed Draft Guidelines for the Production, Processing, Labelling and Marketing of Organically Produced Foods (at Step 7);
 - Proposed Draft Guidelines for the Use of Health and Nutrition Claims (at Step 4);
 - Recommendations for the Labelling of Potential Allergens (at Step 4);
 - Proposed Draft Amendments to the Codex General Guidelines on Claims on the Use of the Term "Natural" (at Step 4);
 - Principles for the Consideration of Nutrient Reference Values for Labelling Purposes
 - Consideration of Labelling as Related to Biotechnology.

DATE AND PLACE OF THE NEXT SESSION (Agenda Item 12)

97. The Committee was informed that its 23rd Session was tentatively scheduled to be held from 24-28 October 1994 in Ottawa, Canada, pending a final decision by the Codex Alimentarius Commission.

SUMMARY STATUS OF WORK

Subject Matter	Step	Action by	Document Reference in ALINORM 93/22
Endorsement of Labelling Provisions in Codex Standards	-	20th CAC CCNFSDU CCFFP 23rd CCFL	paras. 12-18
Section 3.3.4 (Draft NRVs) of the Codex Guidelines on Nutrition Labelling	8	Governments 20th CAC	para. 23 Appendix II
Proposed Draft Guidelines for the Use of Health and Nutrition Claims	3	Governments Canada CCNFSDU 23rd CCFL	para. 48-49 Appendix III
Labelling of Potential Allergens	3	Governments 23rd CCFL	para. 56 Appendix IV
Proposed Draft Guidelines for Organically Produced Foods	5	20th CAC Governments 23rd CCFL	para. 88 Appendix V
Proposed Draft Guidelines for the Use of the Term "Natural"	. 3	Canada Governments 23rd CCFL	para. 95

LIST OF PARTICIPANTS LISTE DES PARTICIPANTS LISTA DE PARTICIPANTES

CHAIRPERSON:

Mrs. Katharine Gourlie

Director

Bureau of Consumer Affairs Consumer Products Branch

Consumer and Corporate Affairs Canada

Place du Portage, Phase 1 Hull, Quebec K1A 0C9

Canada

ARGENTINA ARGENTINE

Mr. Alejandro Meroniuc First Secretary Embassy of Argentina 151 Slater St. # 400 Ottawa, Ontario K1P 5H3

Mr. Guillermo E. Devoto 2nd Secretary Embassy of Argentina 90 Sparks St. # 620 Ottawa, Ontario K1H 2H8

AUSTRALIA AUSTRALIE

Mr. Peter Tough, Director Food Standards Section National Food Authority PO Box 7186 Canberra MC, ACT 2610 Australia

Ms. Ruth Lovisolo
Principal Executive Officer
International Food Standards Policy Section
Australian Quarantine and Inspection Service
GPO Box 858
Canberra ACT 2601
Australia

Mr. Frank Lee Scientific Affairs Manager Goodman Fielder Ltd. PO Box 396 Gladesville, NSW 2111 Australia Dr. Efi Farmakalidis Nutrition and Scientific Research Manager Kellogg (Aust) Pty Ltd 41-51 Wentworth Avenue Pagewood NSW 2019 Australia

CAMEROON CAMEROUN CAMERUN

Mr. Célestin Ngassam Economic Councillor Cameroon Embassy 170 Clemow Avenue Ottawa, Ontario K1S 2B4 Canada

Mr. J.P. Ayafor Fisrt Secretary Cameroon Embassy 170 Clemow Avenue Ottawa, Ontario K1S 2B4 Canada

CANADA

Mr. G. F. Reasbeck Chief, Food Division Consumer Products Branch Consumer & Corporate Affairs Place du Portage, Phase 1, 16th Floor 50 Victoria Street Hull, Quebec K1A 0C9, Canada

Dr. S. W. Gunner Director General - Food Directorate Health Protection Branch Health and Welfare Canada HPB Bldg. Tunney's Pasture Ottawa, Ontario K1A OL2 Dr. M.C. Cheney, Chief Nutrition Evaluation Division Health Protection Branch Health and Welfare Canada Tunney's Pasture Ottawa, Ontario K1A 0L2 Canada

Mr. R. Mills Inspection and Enforcement Fisheries and Oceans Canada 200 Kent Street, 9th Floor Ottawa, Ontario K1A 0E6, Canada

Dr. Ian Sutherland, Director Agri-Food Safety & Strategies Agriculture Canada Halldon House, Room 506 2255 Carling Avenue Ottawa, Ontario K1A 0Y9, Canada

Mr. Roy Harwood Nabisco Brands Canada Association of Canadian Biscuit Manufacturers 885 Don Mills Road Don Mills, Ontario M3C 1V9, Canada

Mr. James Drum
Industry Representative
c/o Coca-Cola Ltd.
1 Concorde Gate, Suite 500
Don Mills, Ontario, Canada

Dr. Suk Hing Yiu Food Industry Competitiveness Branch Ontario Ministry of Agriculture and Food 801 Bay Street, 12th Floor Toronto, Ontario M7A 2B2, Canada

Anne Macey Canadian Organic Growers Box 6408, Station J Ottawa, Ontario K2A 3Y6 Canada

Dale Tulloch Vice President National Dairy Council 221 Laurier East Ottawa, Ontario K1N 6P1

Alison Rimmer
Project Officer
Agriculture Canada
2249 Carling Avenue
Ottawa, Ontario K2B 7E9

Gerald Boutin Conseiller MAPQAQ 200 A Chemin Ste-Foy Quebec, Quebec G1R 4X6

Debra Bryanton
Associate Director
Inspection Strategies
Agriculture Canada
2249 Carling Ave.
Ottawa, Ontario K2B 7E9

Mr. Bernard Dallaire Agri-Food Safety & Strategies Agriculture Canada 2255 Carling Avenue Ottawa, Ontario K1A 0Y9 Canada

Lynn Dolan Labelling Compliance Officer Dept. of Fisheries and Oceans 9th Floor 200 Kent Street Ottawa, Ontario K1A 0E6

J.H. Roy Agriculture Canada 2255 Carling Avenue Ottawa, Ontario K1A 0Y9

Cameron Prince Chief - Fish & Fish Products Fisheries and Oceans Canada 200 Kent Street Ottawa, Ontario K1A 0E6

Jon H. Hutton Associate Director Agri-Food Safety Agriculture Canada 2255 Carling Ave. Ottawa, Ontario K1A 0Y9

Robert Beauchemin President Organic Crop Improvement Association 30 Rte. 214 Milan, Canada GOY 1E0

Dr. Rene R. Riel Technical Adviser Chief Seattle Foundation 60 Therien Hull, Quebec H4X 1Y6 Canada Dr. Gowri Sankaran Everest Swiss Enterprises Ltd. 39 Westminster Avenue South Montreal, Quebec H4X 1Y6 Canada

Mr. Andrew Kawczak Chief Seattle Foudation 39 Westminster Avenue South Montreal, Quebec H4X 1Y6 Canada

Mr. Istvan Mailath Mammuth International Humanitarian Societies 39 Westminster Avenue South Montreal, Quebec H4X 1Y6 Canada

Dr. Farkas Arpad Homonnay Dr. F.A. Homonnay & Associates 39 Westminster Avenue South Montreal, Quebec H4X 1Y6 Canada

Ms. Eunice Chao Sr. Nutrition Scientist Kellogg Canada Inc. 6700 Finch Avenue West Etobicoke, Ontario M9W 5P2 Canada

CUBA

Doris Hernandez Torres Analista Superior de Industria Pesquera Direccion de Calidad Ministerio de la Industria Pesquera Barlovento Santa Fe Playa La Habana Cuba

DENMARK DANEMARK DINAMARCA

Mrs. Anne Busk-Jensen
Deputy Director
Confederation of Danish Industries
H.C. Andersens Boul. 18
Dk-1797 Copenhagen V
Denmark

Dr. Jorgen Fredsted Deputy Director General National Food Agency of Denmark Morkhoj Bygade 19 DK-2860 Soborg, Denmark Kristina Bockhahn Head of Section National Food Agency of Denmark Morkhoj Bygade 19 DK-2860 Soborg - Denmark

FINLAND FINLANDIA

Mrs. Auli Suojanen Senior Food Officer National Food Administration P.O. Box 5 00531 Helsinki - Finland

Dr. Kaija Hasunen
Senior Research Officer Ministry of Social Affairs
and Health
P.O. Box 197
00531 Helsinki
Finland

FRANCE FRANCIA

Madame Roberte Amiel
Chef de délégation
Ministère de l'Economie et des Finances
D.G.C.C.R.F.
59 boulevard Vincent Auriol
75703 Paris CEDEX 13
France

Philippe Mouton Directeur des Affaires Européennes Pernod Ricard Rue de Trèves, 45 1040 Bruxelles, Belgique

Hubert Ferry-Wilczek
Chef du Bureau de la réglementation
Ministere de l'Agriculture et de la Pêche
Direction générale de l'Alimentation
175 Rue de Chevalecet
75013 Paris
France

Catherine Vigreux
Conseiller Reglementaire
ANIA/BSN
7 rue de Teheran
75007 Paris
France

GERMANY ALLEMAGNE ALEMANIA

Prof. Dr. med. Hildegard Przyrembel Scientific Director Federal Health Office Thielallee - Postfach 33 00 13 1000 Berlin 33 - Germany

Mrs. Angelika Mrohs Adviser, Bund für Lebensmittelrecht und Lebensmittelkunde e.V. Godesberger Allee 157 5300 Bonn 2 - Germany

Mr. Friedrich Frede Adviser, Diaetverband e.v. Kelkheimer Strasse 10 6380 Bad Homburg v.d.H. 1 - Germany

GHANA

Mr. Simon S. Puli Minister-Counsellor Ghana High Commission 1 Clemow Avenue Ottawa, Ontario K1S 2A9, Canada

HUNGARY HONGRIE HUNGRIA

Prof. Dr. Péter Biacs Director-General, Central Food Research Institute P.O. Box 393, H-1536 Budapest Budapest II, Herman Otto ut 15 Hungary

ISRAEL

Mr. Kurt Gabor, Director of Food Division Manufacturers Association of Israel c/o Mr. R. Knobil, Secretary Israel Codex Alimentarius Committee Ministry of Industry & Trade P.O.B. 299, Jerusalem, Israel

JAPAN JAPON

Mr. Hidejiro Takizawa Director Office of Health Policy on Newly Developed Foods Environmental Health Bureau Ministry of Health and Welfare 1-2-2 Kasumigaseki Chiyoda-ku Tokyo 100-45, Japan Mr. Yatsuho Ohtani Food Sanitation Specialist Office of Health Policy on Newly Developed Foods Environmental Health Bureau Ministry of Health and Welfare 1-2-2 Kasumigaseki Chiyoda-KU Tokyo, Japan

Shun Yoshioka Technical Advisor Japan Food Hygiene Assoc. 2-6-1 Jingumae Siburya-KU Tokyo, Japan

Mr. Koichiro Hayashida Director, Office of Food Labelling Consumers Economy Division Food and Marketing Bureau Ministry of Agriculture, Forestry and Fisheries Tokyo, Japan

Mr. Kunimasa Koga Technical Advisor Japan Food Hygiene Association 2-6-1 Jingymua Shibuya-Ku Tokyo, Japan

Mr. Yoshio Moriyama Technical Adviser Japan Food Hygiene Association 2-6-1 Jingumae Shibuya-Ku Tokyo, Japan

KUWAIT KOWEIT

Ms. Basma Dashti Kuwait Institute for Scientific Research P.O. Box 2 4885 13109 Safat Kuwait

LITHUANIA LITHUANIE LITUANIA

Dr. Jeronimas Maskeliunas
Director National Nutrition Center, Head of the
Lithuanian Codex Alimentarius Committee
Ministry of Health
Kalvariju 153 - 2042, Vilnius
Lithuania

MEXICO MEXIQUE

Ms. Dolores Contreras Technical Director - Jugos del valle Insurgentes 30 Tepotzotlan, Edo de Mexico Mexico 54600

Mr. Eduardo Cervantes Mexican Delegation Calz. Vallejo 1849 Col. San Jose de la Escalera Mexico, D.F. 07630 Mexico

MICRONESIA (Federated States of) MICRONESIE (Fédération de)

Mr. Nachsa Siren Environmental Health Coordinator Department of Health Services Pohnpei 96941 Federated States of Micronesia

MYANMAR

Daw Aye Aye Myint Second Secretary Embassy of the Union of Myanmar 85 Range Road, Suite 902 Ottawa, Ontario K1N 8J6

Mr. Myint Tun Attache Embassy of the Union of Myanmar 85 Range Road, Suite 902 Ottawa, Ontario K1N 8J6

NETHERLANDS PAYS-BAS PAISES BAJOS

Dr. P.J. Anema Ministry of Welfare, Health and Cultural Affairs Nutrition and Product Safety Affairs P.O. Box 3008 2280 MK Rijswijk (ZH)

A.F. Onneweer
Ministry of Agriculture, Nature Management and
Fisheries
Department for the Environment, Quality and
Nutrition
P.O. Box 20401
2500 EK The Hague, Netherlands

Mrs. I.M.F. Rentenaar General Commodity Board for Arable Products P.O. Box 29739 2502 LS The Hague, Netherlands

G.H. Schipper Commission for the Dutch Food and Agricultural Industry Unilever P.O. Box 760 3000 DK Rotterdam, Netherlands

E. Veen
Commission for the Dutch Food and Agricultural
Industry P.O. Box 93093
2509 AB The Hague, Netherlands

NEW ZEALAND NOUVELLE-ZELANDE NUEVA ZELANDIA

Ms. Elizabeth Aitken Senior Advisor (Nutrition) Food Administration Section Department of Health P.O. Box 5013 Wellington, New Zealand

NORWAY NORVEGE NORUEGA

Ms. Ragnhild Kjelkevik Norwegian Food Control Authority P.O. Box 8187 Dep. N-0034 OSLO, Norway

Mrs. Aase Fulke Norwegian Food Control Authority P.O. Box 8187 Dep. N-0034 OSLO, Norway

RUSSIAN FEDERATION FEDERATION RUSSE FEDERACION RUSA

Mrs. Svetlana Rusanova
Chief Specialist in Scientific and
Technical Department of the Committee of
Fisheries of the Russian Federation
Rozhdestveskly Bul. 15
Moscow, Russian Federation

Mrs. Nina V. Chupakhina Chief, Laboratory of Standardization and Metrology, VNIRO V. Krasnoselskaya 17 Moscow, Russian Federation Mrs. Kira M. Mikhlina Scientist, Centre Mariculture-VNIRO V. Krasnoselskaya 17 Moscow, Russian Federation

SINGAPORE SINGAPOUR

Mr. Sin-I Chu Chief Food Officer Food Control Department Ministry of the Environment 40 Scots Road # 19-00 Singapore 0922 Republic of Singapore

SPAIN ESPAGNE ESPANA

Ms. Dolores Loureda Mantinan Consejero Comercial Ministerio de Industria Embajada de Espana 151 Slater Street - Suite 801 Ottawa, Ontario K1P 5H3

SWEDEN SUEDE SUECIA

Mrs. Eva Lönberg Codex Coordinator National Food Administration - Box 622 S-75126 Uppsala, Sweden

Mrs. Birgitta Lund Senior Administrative Officer Food Standards Division National Food Administration Box 171622 S-75126 Uppsala, Sweden

Mr. Gunnar Rundgren Managing Director KRAV-kontrollforeningen for Ekologisk Odling Brogarden, Jalla S-75594 Uppsala - Sweden

SWITZERLAND SUISSE SUIZA

Mr. P. Rossier Head of Codex Section Federal Office of Public Health Haslerstrasse 16 CH. 3000 Berne 14, Switzerland Mrs. G. Humbert Jacobs Suchard SA CH-2003 Neuchatel Switzerland

Dr. J. Kruseman Nestec SA CH-1800 Vevey Switzerland

Dr. M. Imfeld Hoffmann-La Roche AG CH-4002 Basel - Switzerland

Dr. B. Schmidli Hoffmann-La Roche AG CH-4002 Basel - Switzerland

THAILAND THAILANDE TAILANDIA

Mrs. Chantana Jutiteparak Deputy Secretary - General Food and Drug Administration Ministry of Public Health Bangkok 10200 - Thailand

Mrs. Narumol Gomolsevin Acting Director of Food Control Division Food and Drug Administration Ministry of Public Health Bangkok 10200 - Thailand

Mrs. Pensri Songdej FDA Liaison Manager Diethelm & Co. Ltd. 50/103 SOI Senanikom 1 Paholyothin Rd. Bangkok 10230 Thailand

Mrs. Kusumaln Trivitayanurak Regulatory Affairs and Govt. Public Relation Warner-Lambert Thailand Co., Ltd. 415 Sukumivit Rd. Sumrongnua, Samuthprakarn 10270 Thailand

Mr. Pasan Teparak First Secretary Royal Thai Embassy 180 Island Park Drive Ottawa, Ontario K1Y 0A2

UNITED KINGDOM ROYAUME-UNI REINO UNIDO

Mr. C. Cockbill
Head of Consumer Protection Division
Ministry of Agriculture, Fisheries and Food,
Ergon House c/o Nobel House
17 Smith Square - London SW1P 3JR
United Kingdom

Dr. D.H. Buss Head of Nutrition Branch Ministry of Agriculture, Fisheries and Food Nobel House 17 Smith Square London SW1P 3JR, United Kingdom

UNITED STATES OF AMERICA ETATS-UNIS D'AMERIQUE ESTADOS UNIDOS DE AMERICA

Mr. L. Robert Lake
Director, Office of Policy, Planning & Strategic
Initiatives
Center for Food Safety and Applied Nutrition
(HPS-4) - F.D.A. 200 C Street, S.W.
Washington D.C. 20204, U.S.A.

Mr. Ronald Brewington Chief, Labelling Policy & Approval Br. Food Safety and Inspection Service - U.S.D.A. 1255 West End Court Bldg., Room 209C Washington D.C. 20250, U.S.A.

Dr. John E. Vanderveen Director, Office of Plant and Dairy Foods and Beverages - F.D.A. 200 C. Street, S.W. Washington, DC 20204, U.S.A.

Dr. Elizabeth Yetley, Acting Director Office of Special Nutritionals F.D.A. 200 C Street, S.W. Washington D.C. 20204, U.S.A.

Cheryl Wade, Deputy Director Food Labelling Division Food and Safety and Inspection Service US Department of Agriculture 1255 West End Court Bldg. Washington DC 20250, U.S.A.

Patty L. Woodall
Staff Assistant for Codex Alimentarius
Food Safety and Inspection Service
Room 3175, South Building
U.S. Department of Agriculture
14th & Independence Ave., S.W.
Washington D.C. 20250, U.S.A.

Mr. Harold Ricker
Assistant Director, Transportation
and Marketing Division
Agricultural Marketing Service
U.S. Department of Agriculture
14th and Independence Ave., S.W.
Washington DC 20250, U.S.A.

Betsy Hill
Director, Government Affairs and
Regulatory Policy
The Nutrasweet Company
Box 730 - 1751 Lake Cook Road
Deerfield, IL 60015, U.S.A.

Regina Hildwine
Director
Technical Regulatory Affairs National Food
Processors Association
1401 New York Avenue, N.W.
Washington, DC 20005, U.S.A.

Julia Howell Manager, Regulatory Submissions The Coca-Cola Company P.O. Drawer 1734 Atlanta, GA 30301, U.S.A.

Bruce Silverglade
Director, Center for Science in the Public Interest
1875 Connecticut Avenue, N.W.
Washington, DC 20009-5728, U.S.A.

Dr. Thomas Swinford Manager, Regulatory Affairs Mead Johnson Nutritional Group Mead Johnson Research Center 2400 West Lloyd Expressway Evansville, IN 47721-0001 U.S.A.

Dr. Leila G. Saldanha Manager, Nutrition Program Kellogg Company Science and Technology Center 235 Porter Street Battle Creek, MI 49016, U.S.A.

Richard F. D'Aloisio Manager, Worldwide Regulatory Programs Pepsi-Cola International 100 Stevens Avenue Valhalla, NY 10595, U.S.A. Charles J. Eby Senior Consultant Felishman-Hillard (Monsanto) 1301 Connecticut Avenue, NW Washington, DC 20036, U.S.A.

J. Michael Sligh Chairman National Organic Standards Board 209 Oakwood Court Greenville, SC 29607, U.S.A.

William J. Friedman Counsel New Mexico Organic Commodity Commission P.O. Box 3628 Albuquerque, NM 87190-3628 - U.S.A.

INTERNATIONAL ORGANIZATIONS ORGANISATIONS INTERNATIONALES ORGANIZACIONES INTERNACIONALES

ASSOCIATION OF OFFICIAL ANALYTICAL CHEMISTS (A.O.A.C. International)

Regina Hildwine
Director
Technical Regulatory Affairs
National Food Processors Association
1401 New York Avenue, N.W.
Washington, D.C. 20005 U.S.A.

ASSOCIATION OF EUROPEAN CELIAC SOCIETIES (A.O.E.C.S.)

Mrs. Hertha Deutsch Association of European Celiac Societies - AOECS A. Baumgartnerstr. 44/C5/2302 A-1230 Vienna - Austria

CONFEDERATION OF THE FOOD AND DRINK INDUSTRIES OF THE EEC CONFEDERATION DES INDUSTRIES AGROALIMENTAIRES DE LA CEE (C.I.A.A.)

Mrs. Anne Busk-Jensen H.C. Andersens Boul. 18 Dk-1797 Copenhagen V - Denmark

EUROPEAN COMMUNITY COMMUNAUTÉ EUROPÉENNE

Mrs. N. Sauze, Administrator Directorate General Internal Market and Industrial Affairs Commission of the European Communities Rue de la Loi 200 B-1049 Bruxelles, Belgium Bent Mejborn Administrator Council Secretariat of the European Communities Rue de la Loi 170 B-1048 Bruxelles, Belgium

Mr. Peter Fisch Administrator Consumer Policy Service Commission of the European Communities Rue de la Loi 200 B-1049 Bruxelles, Belgium

Mr. A. Scharpe
Principal Administrator
Directorate General Agriculture
Commission of the European Communities
Rue de la Loi 200
B-1049 Bruxelles, Belgium

INTERNATIONAL DAIRY FEDERATION FÉDÉRATION INTERNATIONALE DE LAITERIE FEDERACION INTERNACIONAL DE LECHERIA

Dr. Ph Pittet
Chairman of IDF Labelling Committee
Société des Produits Nestlé
P.O. Box 352
CH-1800 Vevey, Switzerland

Mr. Dale Tulloch (IDF representative) National Dairy Council of Canada 221 Laurier Avenue East Ottawa, Ontario K1N 6P1

INTERNATIONAL FEDERATION OF GROCERY MANUFACTURERS ASSOCIATIONS (I.F.G.M.A.)

Ms. Gloria Brooks-Ray
Director, Regulatory Affairs and Nutrition Sciences
CPC International Inc.
Box 8000, International Plaza
Englewood Cliffs, N.J. 07632, U.S.A.

Mr. Sherwin Gardner
Senior Vice President, Science and Technology
Grocery Manufacturers of America, Inc.
1010 Wisconsin Ave., N.W.
Washington, D.C. 20007, U.S.A.

Ms. Laurie Curry
Director, Technical
Grocery Products Manufacturers of Canada
885 Don Mills Road, Suite 301
Don Mills, Ontario M3C 1V9
Canada

INTERNATIONAL FEDERATION OF ORGANIC AGRICULTURE MOVEMENTS (I.F.O.A.M.)

Otto Schmid IFOAM Standards Committee Research Institute of Biological Agriculture Bernhardsberg CH-4104 Oberwil/BL Switzerland

Mr. Ken Cummins Program Executive IFOAM Accreditation Program 29 Patrick Street Kilkenny, Ireland

INTERNATIONAL LIFE SCIENCES INSTITUTE (I.L.S.I.)

Reginald J. Ohlson Chair FAO/WHO Coordinating Committee International Life Sciences Institute 1126 16th St. N.W. Washington, D.C. 10036, U.S.A.

INTERNATIONAL ORGANIZATION OF CONSUMERS UNIONS ORGANISATION INTERNATIONALE DES UNIONS DE CONSOMMATEURS (I.O.C.U.)

Ms. Carol Williams c/o I.O.C.U. Regional Office for Europe and North America Emmastraat 9, 2595E4 The Hague, The Netherlands

Ms. Diane McCrea Association for Consumer Research 2 Marylebone Road London, NW1 4DF U.K.

INTERNATIONAL SPECIAL DIETARY FOODS INDUSTRIES (I.S.D.I.)

Mr. Johan Ganzevoort Secretary General International Special Dietary Foods Industries 194, rue de Rivoli F - 75001 Paris, France

Joint FAO/WHO Secretariat Secrétariat mixte FAO/OMS Secretaria conjunta FAO/OMS

Ms. Selma H. Doyran Joint FAO/WHO Food Standards Programme Food and Agriculture Organization Via delle Terme di Caracalla 00100 Rome - Italy

Mr. David H. Byron Joint FAO/WHO Food Standards Programme Food and Agriculture Organization Via delle Terme di Caracalla 00100 Rome - Italy

Canadian Secretariat Secrétariat canadien Secretaria canadiense

Mr. R.B. Burke, Deputy Chief
Food Regulatory, International and Interagency
Affairs Division
Health Protection Branch
Health and Welfare Canada
Room 200, HPB Building
Tunney's Pasture
Ottawa, Ontario K1A 0L2
Canada

Mr. Ian B. Campbell
Consumer Products Branch
Consumer and Corporate Affairs Canada
Place du Portage, Phase 1
50 Victoria Street
Hull, Quebec K1A 0C9
Canada

Ms. Christina Zehaluk
Scientific Evaluator
Nutrition Evaluation Division
Bureau of Nutritional Sciences
Health Protection Branch
Tunney's Pasture
Ottawa, Ontario K1A 0L2
Canada

DRAFT NUTRIENT REFERENCE VALUES FOR FOOD LABELLING PURPOSES¹ (At Step 8 of the Procedure)

(Proposal for amendment to Section 3.3.4 of the Codex Guidelines on Nutrition Labelling)

3.3.4 Numerical information on vitamins and minerals should be expressed in metric units and/or as a percentage of the Nutrient Reference Value per 100 g or per 100 ml or per package if the package contains only a single portion. In addition, this information may be given per serving as quantified on the label or per portion provided that the number of portions contained in the package is stated.

In addition, information on protein may also be expressed as percentages of the Nutrient Reference Value.

The following Nutrient Reference Values should be used for labelling purposes in the interests of international standardization and harmonization:

```
Protein
               (g)
                      800²
Vitamin A
               (μg)
                        5³
Vitamin D
               (μg)
Vitamin C
                       60
Thiamine
               (mg)
                        1.4
Riboflavin
               (mg)
                        1.6
                       18³
Niacin
               (mg)
Vitamin B
               (mg)
                      200
Folic acid
               (μg)
Vitamin B<sub>12</sub>
               (μg)
                      800
Calcium
               (mg)
Magnesium
                      300
               (mg)
               (mg)
Iron
                       14
Zinc
                       15
               (mg)
                      150^{3}
Iodine
               (\mu g)
               Value to be established
Copper
               Value to be established
Selenium
```

¹ In order to take into account future scientific developments, future FAO/WHO and other expert recommendations and other relevant information, the list of nutrients and the list nutrient reference values should be kept under review.

² Proposed addition to Section 3.2.7 (Calculation of Nutrients) of the Codex Guidelines on Nutrition Labelling: "For the declaration of β -carotene (provitamin A) the following conversion factor should be used: 1 μ g retinol = 6 μ g β -carotene

³ Nutrient Reference Values for Vitamin D, Niacin and Iodine may not be applicable for countries where national nutrition policies or local conditions provide sufficient allowance to ensure that individual requirements are satisfied. See also section 3.2.4.1 of the Codex Guidelines on Nutrition Labelling.

PROPOSED DRAFT GUIDELINES FOR THE USE OF HEALTH AND NUTRITION CLAIMS (At Step 3 of the Procedure)

1	•	S	C)(0	P	E	

- 1.1 These guidelines relate to the use of nutrition and health claims in food labelling.
- 1.2 These guidelines apply to all foods for which nutrition and health claims are made without prejudice to specific provisions under Codex standards or Guidelines relating to Foods for Special Dietary Uses and Foods for Special Medical Purposes.
- 1.3 These guidelines are intended to supplement the Codex General Guidelines on Claims and do not supersede any prohibitions contained therein.

2. **DEFINITIONS**

- 2.1 <u>Nutrition claim</u> means any representation which states, suggests or implies that a food has particular nutritional properties including but not limited to the energy value, and to the content of protein, fat and carbohydrates, as well as the content of vitamins and minerals.
- 2.1.1 <u>Nutrient content claim</u> is a nutrition claim that describes the level of a nutrient contained in a food.

(Examples: "source of calcium"; "high in fibre and low in fat";)

2.1.2 <u>Comparative claim</u> is a claim that compares the nutrient levels and/or energy value of two or more foods.

(Examples: "less than"; "more than"; "fewer".)

2.1.3 <u>Nutrient function claim</u> is a nutrition claim that sets out in general terms the nutritional consequences for good health of the intake of a particular nutrient.

(Examples: "Calcium aids in the development of strong bones and teeth";

"Protein helps build and repair body tissues";

"Iron is a factor in red blood cell formation";

"Vitamin E protects the fat in body tissues from oxidation"; "Sugars provide a source of quick energy for the body".)]

2.2 <u>Health claim</u> means any representation that states, suggests or implies that a relationship exists between a food or a nutrient or other substance contained in a food and a disease or health-related condition.

¹Examples included for clarification of definitions.

(Examples:

A. Health-related effects on the body attributed to directly to a food or nutrient or substance

"X fish oil lowers serum triglycerides and increases clotting times."

"X bran lowers blood cholesterol levels."

"X vegetable oil is low in saturated fat and will help reduce blood cholesterol levels".

"Contains soluble fibre that lowers blood cholesterol levels."

"Contains sorbitol. Polyols are more slowly absorbed than sugars and decrease the insulin response."

B. <u>Disease prevention attributed to nutrient or substance contained in a food</u>

"X contains soluble fibre which reduces risk of heart disease."

"X is low in saturated fat which reduces risk of heart disease."

C. <u>Disease prevention or health-related effects related to diet</u>

"A low fat diet will reduce risk of cancer. X is a low fat food."

"Saturated fat raises blood cholesterol levels. A diet low in saturated fat will reduce blood cholesterol levels and reduce risk of cardiovascular disease. X is low in saturated fat."

3. **NUTRITION LABELLING**

Any food for which a nutrition or health claim is made should be labelled with a nutrient declaration in accordance with Section 3 of the Codex Guidelines on Nutrition Labelling.

4. NUTRIENT CONTENT CLAIMS

- When a nutrient content claim that is listed in the Table to these Guidelines or a synonymous claim is made, the conditions specified in the Table for that claim should apply.
- Where a food is by its nature low in or free of the nutrient that is the subject of the claim, the term describing the level of the nutrient should not immediately precede the name of the food but should be in the form "a low (naming the nutrient) food" or "a (naming the nutrient)-free food".

5. COMPARATIVE CLAIMS

Comparative claims should be permitted subject to the following conditions and based on the food as sold:

- 5.1 The foods being compared should be different versions of the same food or similar foods.
- The foods being compared should be clearly identified. A statement of the amount of difference in the energy value or nutrient content should be given. The following

information should appear in close proximity to the comparative claim:

- 5.2.1 The amount of difference related to the same quantity, expressed as a percentage, fraction, or an absolute amount. Full details of the comparison should be given.
- The identity of the food(s) to which the food is being compared. The food(s) should be described in such a manner that it (they) can be readily identified by consumers.
- The comparison should be based on a relative difference of at least 25% in the energy value or nutrient content, except for micronutrients where a 10% difference in the NRV would be acceptable, between the compared foods and a minimum absolute difference in the energy value or nutrient content equivalent to the figure defined as "low" or as a "source" in the Table to these Guidelines.
- 5.4 [The use of "reduced" (e.g. light) or "increased" should be restricted to changes of at least 25% of energy or macronutrients or 10% of the NRVs for micronutrients. This should not preclude factual numerical statements about smaller changes.]

6. **NUTRIENT FUNCTION CLAIMS**

- 6.1 Claims relating to the function of a nutrient in the body should be permitted provided the following conditions are fulfilled:
- 6.1.1 The claim is for a generally recognized and accepted action or effect of a nutrient;
- 6.1.2 The claim is to the effect that the nutrient is a factor or an aid in maintaining the structure and functions of the body necessary to normal growth and development and the maintenance of good health and of activity; and
- 6.1.3 The food for which the claim is made should be a significant source of the nutrient in the diet.

7. HEALTH CLAIMS

- 7.1 A health claim that a food or a nutrient or substance contained in a food has a health-related effect on the body [should/should not] be permitted.
- A claim that the consumption or reduced consumption of a food or nutrient or substance contained in a food as part of a total dietary pattern may have an effect on a disease or health-related condition [should/should not] be permitted.

8. CLAIMS RELATED TO DIETARY GUIDELINES OR HEALTHY DIETS

Claims that relate to dietary guidelines or "healthy diets" should be permitted subject to the following conditions:

- 8.1 Only claims related to the pattern of eating contained in dietary guidelines officially recognized by the appropriate national authority.
- Flexibility in the wording of claims is acceptable, provided the claims remain faithful to the pattern of eating outlined in the dietary guidelines.
- 8.3 Claims related to a "healthy diet" or any synonymous term are considered to be claims

about the pattern of eating contained in dietary guidelines and should be consistent with the guidelines.

- Foods which are described as part of a healthy diet, healthy balance, etc., should not be based on selective consideration of one or more aspect of the food. They should be required to satisfy certain minimum criteria for other major nutrients related to dietary guidelines.
- Foods should not be described as "healthy" or be represented in a manner that implies that a food in and of itself will impart health.
- Foods may be described as part of a "healthy diet" provided that the label carries a statement relating the food to the pattern of eating described in the dietary guidelines.

TABLE²

COMPONENT	CLAIM .	CONDITIONS
A.		NOT MORE THAN
Energy	Low	40 kcal (170 kJ) per 100 g (solids) or 20 kcal (80 kJ) per 100 ml (liquids)
Fat	Low	3 g per 100 g (solids) or 1.5 g per 100 ml (liquids)
	Free	0.15 g per 100 g/ml
Saturated Fat	Low	1.5 g per 100 g (solids) or 0.75 g per 100 g (liquids) and 10% of energy
Cholesterol	Low	20 mg per 100 g (solids or 10 mg per 100 ml (liquids) and 1.5 g per 100 g (solids) or 0.75 g per 100 g (liquids) and 10% of energy
Sugars	Free	0.5 g per 100 g/ml
Sodium	Low [Very Low] [Free]	120 mg per 100 g [40 mg per 100 g] [5 mg per 100 g]
В.		NOT LESS THAN
Fibre	Source	[2 g per serving]
	High	[4 g per serving]
Protein	Source	[10% of reference RDA/100 g]
	High	[20% of reference RDA/100 g]
Vitamins and Minerals	Source	[10-15% of reference RDA/100 g]
	High	[20-30% of reference RDA/100 g]

²As amended by Codex Committee on Nutrition and Foods for Special Dietary Uses, their 18th Session September 28 - October 2, 1992.

CONSIDERATION OF POTENTIAL ALLERGENS IN FOODS

(Working paper prepared by Norway, in cooperation with Finland, Iceland and Sweden)

1. TERMS OF REFERENCE AND SCOPE OF THE DOCUMENT

At its 21st Session, the Codex Committee on Food Labelling (CCFL) agreed to discuss a working paper at its next session which would examine the labelling of potential allergens which were included as components of composite ingredients in foods, and thus were not included in the ingredients list. The working paper would be prepared under the direction of Norway, with assistance provided by Finland, Iceland and Sweden.

The Committee noted that this issue would require examination of Section 4.2.1.3 of the General Labelling Standard, which addressed the labelling of composite ingredients, especially as related to the "25% rule". The working paper would examine this issue in detail, and would include possible recommendations to the Committee for their consideration. It was also concluded that the CCEXEC would be informed of CCFL deliberations in this area (ALINORM 91/22, paras. 146-147).

At its thirty-eighth session, the Executive Committee expressed "concern as to the practicality of the proposal, as almost all foods contained potential allergens. With this in mind, the Executive Committee recommended that the CCFL should proceed cautiously when examining this subject" (ALINORM 91/4, paras. 49-50).

The present working paper summarises the main issues of the matter. These are considered in more depth in the annex to the paper, which also contains a list of references.

In accordance with the above terms of reference, the main emphasis of the present document is on problems associated with the declaration of composite ingredients. However, certain other aspects of labelling which are of concern to consumers with hypersensitivity problems have also been addressed, in particular the use of class names.

2. INTRODUCTION

The Codex General Labelling Standard provides the basis for national food labelling regulations in many countries throughout the world.

Because of the widespread occurrence of hypersensitivity problems, often of a serious nature, and of the ever-increasing variety of food products, often unfamiliar, offered to the consumer, the need for more informative labelling to help those affected is also increasing. The Provisions of the General Standard should reflect this situation.

3. HYPERSENSITIVITY - ALLERGY AND INTOLERANCE

Hypersensitivity reactions can be divided into allergy, which is a condition in which the immune response is altered, and intolerance for which no immunological cause has been detected. Intolerance may arise in association with metabolic disorders, such as lactose intolerance, or may be of unknown etiology. Coeliac disease involves the immunesystem but is not an allergic disease. The mechanism is not fully known.

Common allergens are proteins, whereas a variety of substances may cause intolerance.

In this document hypersensitivity is used as a collective term to denote both allergy and intolerance.

3.1 Occurrence

Data and estimates concerning the occurrence of hypersensitivity in general vary considerably from just a few to more than 40% of the population. Even the most conservative estimates imply that a large number of consumers are affected by hypersensitivity in some form or another. If only 2% were actually to suffer from such reactions, which is probably an underestimation, this would mean millions of affected consumers in Europe alone.

A Swedish official report concludes that food hypersensitivity occurs in about 20% of children up to three years old, 8% among six-year olds, about 15% among young adults, and is probably just as common in older people.

Hypersensitivity to food additives is uncommon in people who are not otherwise affected by allergic conditions. A Danish study reports a prevalence of 1-2% among school children.

Lactose intolerance is very common on a world basis, in some countries 90% of the population being reported to have this intolerance.

Coeliac disease has also a very high occurrence in some countries, for example in Sweden, where 1 in 300 are reported to suffer from the condition. The prevalence of both types of intolerance, however, varies considerably from country to country.

A reasonable estimate of the overall incidence of food hypersensitivity in some form would seem to be about 10% of the population.

3.2 Food which most commonly give rise to hypersensitivity

A number of commmon foods cause symptoms in many people in many parts of the world, which may be unpleasant and reduce the quality of life. Other allergies and intolerances are not so common, but nevertheless very serious for those concerned. Normal foods can induce reactions of a crippling, even life-threatening, nature.

It is difficult to make a complete and exhaustive list because hypersensitivity reactions are very individual, and because hypersensitivity reactions are related to patterns of consumption and dietary habits, especially in childhood. For instance, allergy to rice in Japan and to corn (maize) and peanuts in USA is more common in these countries than in Scandinavia. This means that there will be differences between population groups and countries as regards the foods most commonly giving rise to allergy and intolerance.

The foods and ingredients which have been well-documented and generally accepted to give rise to hypersensitivity reactions in people are: cereals; eggs, fish; legumes; milk; nuts, almonds etc.; fish; shellfish/crustacea. These foods may cause manifest symptoms in a large

number of people, and may be life-threatening or cause long-standing medical problems in small amounts in susceptible people.

Several types of fruit and vegetable give rise to symptoms in people with pollen allergies. Aspects such as amount, severity, effects of processing etc., have, however, not been fully elucidated.

As regards food additives, it has been documented that sulphites at levels down to 10 ppm may cause severe reactions in susceptible individuals. It is also known that colours, especially azo-colours, preservatives such as benzoates, and antioxidants such as BHA and BHT, give rise to hypersensitivity. The effects of these and other substances, especially concerning the amounts necessary to trigger off reactions, appear not to have been fully clarified.

4. PROBLEMS ASSOCIATED WITH LABELLING REGULATIONS

The essential requirements which hypersensitive consumers have as regards food labelling are:

- a) that the declaration of ingredients must be sufficiently precise so as to enable the avoidance of food components which are not tolerated, and
- b) that the labelling designations used must be familiar, unequivocal, and not subject to misunderstanding.

The main rule for the declaration of ingredients in the Codex General Standard on Labelling as contained in Section 4.2.1.2 is that there shall be a complete and detailed declaration.

However, less specific declaration of ingredients is permissible in certain cases. Especially important in this connection are the so-called 25% rule (4.2.1.3), and the use of class names (4.2.2.1).

4.1. Compound ingredients - the 25%-rule

Section 4.2.1.3 of the General Standard reads:

"Where an ingredient is itself the product of two or more ingredients, such a compound ingredient may be declared, as such, in the list of ingredients provided that it is immediately accompanied by a list in brackets of its ingredients in descending order of proportion (m/m). Where a compound ingredient for which a name has been established in a Codex standard or in national legislation constitutes less than 25% of the food, the ingredients other than food additives which serve a technological function in the finished product need not be declared."

The 25% rule means that a substance which is present in only a small amount as an ingredient in a food has to be declared, while the same substance present in fairly large amounts as a component of a compound ingredient need not be declared.

According to the General Standard, the rule shall only be applied to products for which there is a Codex standard, or which are defined in the national legislation of the country in which the food is sold. This must mean that the compound ingredients which are not declared in full as to their component ingredients, must have a composition which is the same at all times in all products. This also presumes that the composition of the ingredient is familiar to the consumer, or that the necessary information is readily available.

The current trend in food legislation policy away from vertical commodity standards and towards general horizontal legislation with emphasis on labelling, implies the presence on the market of an increasing number of products and ingredients without the standardised composition which is the prerequisite for being able to omit the declaration of the components of a compound ingredient according to the 25% rule.

The 25% rule is in many countries also being applied to products which are produced according to custom or tradition, without the composition necessarily being specified in a standard.

Increasing international trade and greater efforts to achieve international harmonisation of food regulations, has resulted in a tendency for countries to accept products which are in accordance with other countries' standards or traditions. If these are included as compound ingredients in another product, there is every likelihood that specification of these compound ingredients will not be required. Consumers will usually not be aware of the standards or customs in other countries; nor will it be easy to obtain the relevant information.

In practice many countries apply the rule to <u>all</u> compound ingredients present at a level of less than 25%, and not just those for which there is a compositional standard.

Extensive world trade in foods, and developments in processing and technology increase these problems.

As many processed foods, non-traditional ingredients, imported foods with a non-traditional composition, and new and novel ingredients are unfamiliar to the consumers, it is essential that the labelling is sufficiently detailed.

The unsatisfactory declaration of compound ingredients concerns not only the rule per se, but also the way in which this Codex provision is interpreted in different countries.

This situation gives cause to raise the question as to whether the 25% rule is in fact still a valid one. A rule which in many parts of the world is not being implemented as was intended, and which is no longer in harmony with current practice, has perhaps outplayed its role.

4.2 Class names

The use of class names may conceal the presence of allergenic ingredients. This is an unsatisfactory situation for hypersensitive people, and may result in unexpected hypersensitivity reactions, or at least cause such consumers to avoid more foods than actually necessary.

4.3 Food additives and processing aids

Food additives carried over with raw materials/ingredients at levels which do not serve a technological function in the finished product, and processing aids, need not be declared.

As regards food additives and hypersensitivity, there is evidence that sulphites induce hypersensitivity reactions when present at levels exceeding 10 mg/kg, i.e. below the level for technological effect.

The significance of other potential intolerance-inducing food additives (e.g. benzoates, azo-colours, antioxidants, glutamate etc.) being present without declaration is less certain and needs further investigation. It is also a question as to how the carry over-principle and the term "technological effect" is interpreted and practised.

4.4 Conclusion

The General Labelling Standard is not satisfactory as regards ensuring that hypersensitive people receive adequate information to enable them to limit their selection of foods to those which they can tolerate, and help them avoid those which they cannot. Because of the extent and gravity of the problem, it seems proper that the consumers concerned should receive the information they need concerning the composition of foods, either through improved labelling regulations, or other appropriate means.

5. ALTERNATIVES TO LABEL DECLARATIONS

Detailed product information could be collated in data banks, product catalogues, etc. However, there will always be uncertainty as to whether such information is up-to-date.

As reliance on wrong or incomplete information may have considerable adverse health implications, such systems may in fact aggravate the situation for the consumers concerned, unless they function properly. This type of information is unlikely to cover the whole range of products, it is unlikely to reach out to all consumers, and it is unlikely that the products included in catalogues are always available to all consumers.

Product catalogues, databases etc would be useful supplements to label information, but could not replace detailed labelling.

Nor is it possible to develop symbols or the like to adequately replace detailed and correct labelling. This idea must be treated with great caution. People suffering from hypersensitivity may react to several food components, the combinations of specific allergies/intolerances varying from person to person. A product which is suitable for one patient may be life-threatening for another.

6. IMPROVEMENT OF LABELLING REQUIREMENTS IN RELATION TO HYPERSENSITIVITY

The only satisfactory solution with regard to ensuring that consumers suffering from hypersensitivity conditions are given adequate information, is to improve labelling requirements.

Incomplete labelling may result in the ingestion of food which may cause considerable discomfort, and at worst serious illness and even death.

The key principle with regard to labelling provisions in relation to allergy and intolerance should be that ingredients which may induce hypersensitivity should, as far as possible, be specifically mentioned in the list of ingredients. Particular emphasis must be given to substances of which even small amounts may trigger off severe reactions, such that ingredients which give severe, debilitating or life-threatening symptoms are always declared.

6.1 Amendment of the 25%-rule

As regards the 25% rule, an appropriate solution would seem to be to reduce the lower limit above which declaration of the components of a compound ingredient is required. Based on an assessment of what is feasible, this lower limit could be set at 5%. Such a limit would necessitate full declaration of the ingredients in a compound ingredient in very many cases. However, even this lower limit will not ensure the declaration of potent allergens capable of

inducing severe hypersensitivity reactions in small amounts. Should compound ingredients contain such components, measures should be taken to make sure that these are always declared.

6.2 Amendment of the provisions concerning class names

Although the ideal solution (from the point of view of hypersensitivity) would be to revoke the provisions allowing declaration of ingredients by class names, this seems to be unrealistic, at least in the short term. However, it should be required that at least ingredients containing the most potent substances be always declared specifically, and not just by class names.

6.3 Amendment to the rules for the declaration of carried-over food additives

Food additives which are carried over with raw materials and compound ingredients, and which can induce hypersensitivity reactions in amounts present in the product, should always be declared regardless of whether they have technological effect or not. This has been documented for sulphites, and the corresponding situation for other additives and processing aids should be investigated.

6.4 Declaration of potent hypersensitivity-inducing substances

Provisions in the General Labelling Standard concerning the declaration of compound ingredients, class names, and the declaration of food additives carried over with raw materials/ingredients, should be amended so as to always require the specific declaration of ingredients which may cause severe allergic or hypersensitivity reactions. A list of ingredients which should always be declared by specific name should be elaborated. The list should be open-ended and be supplemented as needed, in the light of new information.

7. RECOMMENDATIONS

7.1 Amendment of the Codex General Labelling Standard

In light of the considerations presented in this document, it is recommended that the Committee (CCFL) discuss the following proposed amendments to the Codex General Labelling Standard (text to be deleted in brackets) text to be added underlined:

Section 4.2.1.3:

Where an ingredient is itself the product of two or more ingredients, such a compound ingredient may be declared, as such, in the list of ingredients, provided that it is immediately accompanied by a list in brackets of its ingredients in descending order of proportion (m/m). Where a compound ingredient (for which a name has been established in a Codex standard or in national legislation) constitutes less than (25%) 5% of the food, the ingredients other than food additives which serve a technological function in the finished product and ingredients known to cause allergic or intolerance reactions, need not be declared.

The following foods and ingredients are known to cause hypersensitivity and shall always be declared as such:

Barley, oats, wheat, rye, triticale, and products of these (gluten and starch included)

Crustaceans, shellfish, and products of these

Eggs and egg products

Fish and fish products

Legumes: peas, peanuts, soybeans and products of these

Milk and milk products (lactose included)

Sulphite in concentrations of 10 mg/kg or more

Tree nuts, poppy seeds, sesame seeds and products of these

Section 4.2.2.1:

Except for ingredients known to cause reactions of allergy or intolerance such as those listed in Section 4.2.1.3, the following class names may be used.....etc. (rest of section as is).

Section 4.2.3.2:

A food additive carried over into foods at a level less than that required to achieve a technological function, and processing aids, are exempted from declaration in the list of ingredients. The exemption does not apply to food additives and processing aids known to cause reactions of food allergy or intolerance, such as those listed in in Section 4.2.1.3.

7.2 Further consideration of the issues.

Carried-over food additives and processing aids.

In addition to the above recomendations, it is proposed that the Codex Committee on Food Additives and Contaminants (CCFAC) is requested to assist in assessing labelling requirements for carried-over food additives and processing aids.

CCFAC should be asked to report on what hypersensitivity-inducing food additives and processing aids are likely to be present in a product in amounts which may induce a hypersensitivity response, without, under current requirements, having to be declared. As regards carried-over food additives, the relationship between levels which have a technological function and levels which are required to elicit a reaction will be important. It is also of importance to define precisely how the term "technological function" is to be understood and interpreted in practice, and to consider whether declaration of intolerance-inducing food additives should be required above a certain level rather than being tied to the concept of technological effect.

Other ingredients

Recommendations concerning other ingredients such as rice, corn (maize), oils, fruits, vegetables, spices, hydrolized proteins etc. require further investigation.

Annex

LABELLING OF FOODS THAT MAY CAUSE HYPERSENSITIVITY.

Background document

Annex to the working document "Labelling of potential allergens"

Prepared by Norway (Ragnhild Kjelkevik) in cooperation with Finland, Iceland and Sweden.

CONTENTS:

- BACKGROUND
- HYPERSENSITIVITY REACTIONS
 - 2.1 Allergy
 - 2.2 Intolerance/pseudo-allergy
 - 2.4 Occurrence
 - 2.5 Foods which most commonly give rise to hypersensitivity
- 3. PROBLEMS ASSOCIATED WITH LABELLING REGULATIONS
 - 3.1 Compound ingredients the 25% rule.
 - 3.1.1 Prerequisites for the application of the 25% rule
 - 3.1.2 Implementation of the 25% rule
 - 3.1.3 Evaluation of the 25%-rule
 - 3.2 Ingredient designations
 - 3.3 Class names
 - 3.4 Declaration of carried-over food additives and processing aids
 - 3.5 Conclusion
- 4. POSSIBLE ALTERNATIVES TO THE LABEL DECLARATION OF INGREDIENTS
 - 4.1 Product catalogues, data banks etc.
 - 4.2 Symbols and logos
- 5. IMPROVEMENT OF LABELLING REQUIREMENTS
 - 5.1 The 25%-rule
 - 5.2 Class names
 - 5.3 Declaration of carried-over food additives
 - 5.4 Declaration of potent hypersensitivity-inducing substances

6. RECOMMENDATIONS

- 6.1 Amendment of the Codex General Labelling Standard
- 6.2 Further consideration of the issues.

7. LIST OF REFERENCES

1. BACKGROUND

The Codex General Labelling Standard provides the basis for national food labelling regulations in many countries throughout the world. In light of the widespread occurrence of hypersensitivity problems, and of their not infrequent serious nature, and of the consequent critical significance of food labelling for people with allergies, there seems to be good reason to re-examine the Standard with the needs of this group of consumers in mind.

Various forms of food hypersensitivity affect many people, either directly, or indirectly (for example when responsible for preparing food for family members with allergies). Hypersensitivity in general is becoming more common (1,2). Lacking valid data for prevalence, it is not known whether the occurrence of hypersensitivity to food in general is increasing. The frequency, of allergy to pollen is, however, rising, which could mean that the prevalence of pollen-related reactions to foods like nuts, apples etc is also increasing.

The food components causing the hypersensitivity varies with the individual.

Hypersensitivty reactions also vary considerably both as to type and degree of severity, from mild discomfort to life-threatening symptoms. However, generally speaking, hypersensitivity gives rise to problems, sometimes serious, in the daily life of many consumers. The only effective countermeasure is to avoid the substances which connot be tolerated.

Problems are mostly due to commonly used foodstuffs and ingredients, foodstuffs which are desirable in the diet of normal healthy people. It is not feasible to protect hypersensitive consumers by forbidding or limiting the use of such foodstuffs. The regulation of food additives in this connection would only have a marginal effect, and would have to be limited to apply to certain types of foods, for example staple foods.

Labelling constitutes the most important aid in enabling the consumer to avoid products likely to induce hypersensitivity reactions. It is therefore essential that potentially hypersensitivity-inducing ingredients are properly declared.

The current trend is towards the production of more and more processed compound foods, using new and non-traditional ingredients. The horizontal approach to food law has resulted in a move away from commodity standards towards general regulations with emphasis on informative labelling. Foods with a standardised composition are becoming less common.

The removal of trade barriers and the freer movement of foods in international trade, also gives rise to a situation where consumers are faced with new and unfamiliar products. This adds importance to the international harmonisation and uniform application of labelling regulations.

Consumers have less insight into the composition of the variety of food products on the market, and are therefore dependent upon adequate information in order to know what they are buying

2. HYPERSENSITIVITY REACTIONS

Hypersensitivity reactions are classified somewhat differently in different parts of the world. The most important distinction is between allergy, and what can be termed intolerance/pseudo-allergy.

In this document hypersensitivity is used as a collective term to denote both allergy and intolerance.

2.1 Allergy

Allergy is a condition in which the immune response is altered. The response is specific and can be induced by very small amounts of the substance in question, the allergen.

Well documented common allergens are proteins from milk, eggs, fish, nuts, legumes, and cereals. For some people, minute amounts of these allergens can give rise to life-threatening symptoms.

Allergies to proteins in fruits and vegetables are common in patients suffering from pollen allergy (hay fever). These are probably caused by sensitisation to pollen via the respiratory tract. IgE antibodies cross-react with epitopes in pollen and in the fruit or vegetables in question (3).

The possibility of certain contact allergens such as nickel being able to induce reactions on ingestion has also been suggested (4).

2.2 Intolerance/pseudo-allergy

It is sometimes impossible to determine an immunological cause of hypersensitivity reactions. In such cases, the condition is often termed intolerance or pseudo-allergy.

Hypersensitivity is also associated with metabolic disorders, for example lactose intolerance. Other reactions are of unknown etiology. Coeliacic diseases involves the immunesystem but is not an allergic disease. The mechanism is not fully known.

The symptoms of intolerance may be the same as observed in food allergy.

Proteins may also cause allergy-like symptoms without any role of the immune system being identified.

Certain food additives have been stated to cause intolerance.

2.3 Symptoms

Symptoms of allergy manifest themselves not only in the alimentary tract but also in the skin, respiratory tract, or as general systematic signs:

Gastro-intestinal tract

Itching and swelling of the mouth, nausea, vomiting, colic,

diarrhoea

Skin

Urticaria, eczema

Respiratory tract

Rhinitis, asthma

Systematic

Anaphylactic shock

2.4 Occurrence

With regard to children, studies carried out in Finland (5) and USA (6) have revealed that approx. 8% of children under, respectively, three and four years of age showed confirmed hypersensitivity reactions.

Some authors estimate the incidence of food allergies in children to be 1-3%, with a figure perhaps as high as 30% in some groups, for example, children with atopic dermatitis (7), while others consider that approx. 20% of all children react (8),

The frequency of many typical child allergies falls with increasing age. Other types of hypersensitive response, however, arise later on, especially the food allergies which are associated with pollen allergy. Approx. 50% of patients with birch pollen allergy react to food, especially nuts, and fruits and vegetables (8). This type of allergy is probably on the increase as respiratory tract allergies are becoming more common.

Because it is difficult to perform satisfactory epidemiological studies of the prevalence of hypersensitivity to foods in adults, few such studies have been carried out.

A Swedish study indicates that about 15% of people between 25-30 years old have some form of food hypersensitivity (8). Others refer to estimates in the area of 10-15% (9,10).

A Swedish official report mentions that the prevalence data to be found in the literature varies considerably, from a few percent up to 43%, but concludes that food hypersensitivity occurs in about 20% of children up to three years old, 8% among six-year olds, about 15% among young adults, and is probably just as common in older people (8)

Hypersensitivity to food additives is uncommon in people who are otherwise not affected by allergic conditions. Investigations in England showed that though about 7% of the population considered themselves to be hypersensitive to food additives, this could only be confirmed in 0.2% at the most (11).

Studies in Denmark revealed a prevalence in schoolchildren of approx.1-2%. The children reacted to food additives such as artificial colours and preservatives. All the children who reacted to food additives had some form of concurrent allergic condition (12).

A controlled challenge of asthma patients who stated that they did not tolerate red wine showed that 5% reacted to sulphite in the red wine (13). In another investigation (14) of asthmatics, it was shown that about 5% of the patients reacted to sulphite (15), tartrazine and other food additives. Patients suffering from urticaria may react to tartrazin and other azo-colours (15b).

As regards specific diseases, it should be mentioned that coeliac disease is considered to affect at least one European in a thousand. However, there is considerable inter-country variation. In certain European countries, the frequency seems to be increasing, being as high as one in 300 in Sweden (16). Frequency of one in 496, and one in 585, have been reported in Austria (17) and Israel (18), respectively.

Lactose intolerance is very common on a world basis, prevalence being very high among certain population groups in some countries. For example, figures of up to 87-89% of people with lactose intolerance have been reported from several Asiatic and African countries. In other countries, for example among the dark-skinned population in North America, India, and Israel, prevalence can be about 50%, compared with the corresponding figure of 3% in Denmark (19).

Data and estimates concerning frequency of food hypersensitivity in general show considerable variation, from a few percent to about 40% of the population. However, even the most conservative figures imply that a large number of consumers are affected by hypersensitivity in one form or another. If only 2% were actually to suffer from such reactions, which is probably an underestimation, this would mean millions of affected consumers in Europe alone.

A reasonable estimate for the overall incidence of food hypersensitivity would seem to be about 10%.

2.5 Foods which most commonly give rise to hypersensitivity

A number of common foods cause symptoms in many people in many parts of the world, which may be unpleasant and reduce the quality of life. Other allergies and intolerances are not so common, but nevertheless very serious for those concerned. Normal foods can induce reactions of a crippling, even life-threatening, nature.

A brief account is given in the following of the foodstuffs and ingredients which most often induce reactions. Many of these foods, even in small amounts, may trigger off anaphylactic shock in susceptible people.

The list is not complete or exhaustive, both because hypersensitivity reactions are very individual, and because hypersensitivity reactions are related to patterns of consumption and dietary habits, especially in childhood. For instance, allergy to rice in Japan and to corn (maize) and peanuts in USA is more common in these countries than in Scandinavia. This means that there will be differences between population groups and countries as regards the foods most commonly giving rise to allergy and intolerance.

Nuts, almonds, and the like: People allergic to nuts may react to very small amounts, reactions such as anaphylactic shock not being unusual. People who are allergic to pollen from deciduous trees often suffer from allergy to nuts.

Fish: Fish allergy is not among the most common forms of allergy, but often give rise to dramatic symptoms, - for some people minute amounts of fish is enough to cause an attack. All species of fish may be involved. People with egg allergy sometimes also have fish allergy.

Shellfish and crustacea: It is not uncommon for such foods to give rise to pure allergic reactions and to other forms of hypersensitivity.

Eggs: Egg allergy is most frequently seen in children. Most of the allergenic substances are present in the egg white.

Milk: Cows-milk allergy with acute symptoms may occur in infancy, perhaps immediately after weaning. The allergy may be transient, most three to four-year olds tolerating milk. Another type of milk allergy is the so-called slow form, which often affects somewhat older children.

Milk may also induce intolerance reactions. Many people in the world lack lactase, the enzyme breaking down lactose.

Legumes: Proteins in soy beans, peas, peanuts, and other legumes, often cause allergic reactions. Such allergies may be life-threatening, for example, soy, peanut and pea allergies.

Protein residues in, for example, starch and fibre products from these sources may be enough to give cause manifest symptoms of hypersensitivity.

Cereals: Cereals may give rise to several types of hypersensitivity. Wheat, rye, barley, and oats, contain the protein gluten, which damages the intestinal mucosa in some people (coeliac disease). Very small amounts of gluten are sufficient to cause illness in people suffering this condition.

Cereais may also cause allergy.

Fruit and vegetables: This members of this large and heterogenous group of foods contain substances which may cause reactions. The substances giving rise to symptoms are often heat-labile. As mentioned previously, many of the foods in this group give cross-reactions with pollen. Reactions may be severe.

Food additives: In spite of the widespread view that many food additives can cause hypersensitivity reactions, this has been actually confirmed for only a very few substances. These are colours (mainly azo-colours), the preservatives benzoic acid/benzoates and sulphites, and the antioxidants butylhydroxyanisol (BHA) and butylhydroxytoluene (BHT) (20,21).

As mentioned above, hypersensitivity to food additives is very unusual in people not already suffering from an allergic condition. Symptoms in some cases may be grave; this is especially the case with regard to reactions against sulphite in people suffering from asthma, amounts of sulphite as low as 10 ppm giving symptoms (22).

It is otherwise difficult to find unequivocal data in the literature regarding the amounts of various food additives which can give rise to reactions. However, it seems that in some cases small amounts may be involved, 50 mg og benzoate having, for example, been shown to cause reactions (23). The same is true for BHA/BHT (24) and azo-colours (25).

Although glutamate has been described as a cause of asthma, it seems mainly to have other modes action (26).

As regards food additives, reference is also made to a background document elaborated for the 1985 meeting of the (then) Codex Committee on Food Additives (27).

The foods and ingredients which have been well-documented and generally accepted to give rise to hypersensitivity reactions in people are: cereals; eggs; fish; legumes; milk; nuts, almonds etc.; shellfish/crustacea.

These foods may cause manifest symptoms in a large number of people, and may be life-threatening in small amounts in susceptible people. Several types of fruit and vegetable give rise to symptoms in people with pollen allergies. Aspects such as amount, severity, effects of processing etc., have, however, not been fully elucidated.

As regards food additives, it has been documented that sulphites at levels down to 10 ppm may cause severe reactions in susceptible individuals. It is also known that colours, especially azo-colours, preservatives such as benzoates, and antioxidants such as BHA and BHT, give rise to hypersensitivity. The effects of these and other substances, especially concerning the amounts necessary to trigger off reactions, appear not to have been fully clarified.

3. PROBLEMS ASSOCIATED WITH LABELLING REGULATIONS

The essential requirements which hypersensitive consumers have as regards food labelling are:

- a) that the declaration of ingredients must be sufficiently precise so as to enable the avoidance of food components which are not tolerated, and
- b) that the labelling designations used must be familiar, unequivocal, and not subject to misunderstanding.

The main rule for the declaration of ingredients in the Codex General Standard on Labelling is contained in Section 4.2.1.2.

"All ingredients shall be listed in descending order of ingoing weight(m/m) at the time of the manufacture of the food."

In other words, the main rule is that there shall be a complete and detailed declaration.

However, less specific declaration of ingredients is permissible in certain cases. It is these exemptions i.e. the so-called 25% rule, and declaration by class names, which cause problems for hypersensitive consumers, and which therefore must be examined with regard to possible amendment to alleviate the situation.

3.1 Compound ingredients - the 25% rule.

Section 4.2.1.3 of the General Standard reads:

"Where an ingredient is itself the product of two or more ingredients, such a compound ingredient may be declared, as such, in the list of ingredients provided that it is immediately accompanied by a list in brackets of its ingredients in descending order of proportion (m/m). Where a compound ingredient for which a name has been established in a Codex standard or in national legislation constitutes less than 25% of the food, the ingredients other than food additives which serve a technological function in the finished product need not be declared."

This provision leads to the paradoxical situation that a substance added to a food in small amounts as an ingredient must be declared, whereas the same substance can be present in much larger amounts in another product, as a component of a compound ingredient, without being declared.

3.1.1 Prerequisites for the application of the 25% rule

According to the General Standard, the rule shall only be applied to products for which there is a Codex standard, or which are defined in the national legislation of the country in which the food is sold. This must mean that the compound ingredients which are not declared in full as to their component ingredients, must have a composition which is the same at all times in all products. This also presumes that the composition of the ingredient is familiar to the consumer, or that the necessary information is readily available.

There is every reason to question whether these presumptions are fulfilled for foods on the market today. Moreover, commodity standards do not always regulate <u>all</u> ingredients in a food product, which means that ingredients can be present which are not specified in the standard.

The most important consideration, nevertheless, is that both in Codex and at the national level, work on detailed recipe standards is being played down in favour of general, horizontal rules, with stress on, among other things, informative labelling.

The current trend in legislative policy implies that more and more food products and ingredients are appearing on the market which do not have the standardised composition on which the 25% rule is based. In theory, the rule should then not be applied.

3.1.2 Implementation of the 25% rule

In many countries the 25% rule is also being applied to products which are produced according to custom or tradition, without the composition necessarily being specified in a standard. In practice, this often mean most or all composite ingredients present in amounts less than 25%.

Increasing international trade and greater efforts to achieve international harmonisation of food regulations, has resulted in a tendency for countries to accept products which are in accordance with other countries' standards or traditions. If these are included as compound ingredients in another product, there is every likelihood that the declaration of the component ingredients in the compound ingredient will not be required.

Consumers will usually not be aware of the standards or customs in other countries; nor will it be easy to obtain the relevant information.

There thus seems to be trend that the rule is applied to <u>all</u> compound ingredients present in the product at a level of less then 25%.

3.1.3 Evaluation of the 25%-rule

The 25%-rule creates difficulties for people who for health reasons must avoid eating a number of commonly used food ingredients.

Extensive world trade in foods and developments in processing and technology increase these problems.

As many processed foods, non-traditional ingredients, imported foods with a non-traditional composition, and new and novel ingredients are un-familiar to the consumers it is essential that the labelling is sufficiently detailed.

For instance, fat substitutes made from egg and milk protein have been shown to induce symptoms in people allergic to eggs and milk (28). It is essential that ingredients like these are specified on the label.

It can be claimed that the basis for the 25%-rule has been considerably undermined by the fall in the number of standardised products, the increasing volume of international trade in food, and the fact that foods and ingredients with different compositions are being marketed under similar names.

This unsatisfactory declaration of compound ingredients concerns not only the rule per se, but also the way in which this Codex provision is interpreted in different countries.

This situation gives cause to raise the question as to whether the 25%-rule is in fact workable in practice. A rule which in many parts of the world is not practised as intended, and which is difficult to adapt to the current situation, has outplayed its role.

3.2 Ingredient designations

There is no doubt that the most important aspect of labelling as regards hypersensitivity is to ensure adequate specification of the list of ingredients. However, it must be emphasised that it is also very important that ingredient designations are unequivocal and not open to misunderstanding.

Increasing international trade will aggravate the problem in that designations might well be used that are understandable in the exporting country but which do not provide enough information to consumers in the importing country.

It is also important that the ingredient declaration is sufficiently specific.

3.3 Class names

Section 4.4.2.1 of the General Standard provides for the declaration of certain ingredients by class name instead of specific name. Several of these classes include ingredients which may induce reactions in hypersensitive people.

For example, peanut oil is in some countries associated with allergic reactions, but need not be declared specifically, only by the class name "vegetable oil". The same may be the case for other fats and oils.

Starchs from various sources can be declared as "starch" without specification of source. There is a possibility that pea starch may contain sufficient amounts of pea protein to trigger off an allergic reaction in pea-hypersensitive people. The label declaration in this case will give no indication that the product contains pea components. Apparently some kinds of wheat starch may contain gluten in amounts large enough to affect consumers with coeliac disease.

As regards spices, the situation is unclear. There are some indications of sensitive people reacting to nutmeg in amounts less than 2%, when only the class name "spice" is required.

Consumers who are aware that they do not tolerate certain ingredients within a class will often "play safe" and simply not buy a product, when the label does not give a specific indication of the ingredients.

In light of the above considerations, the use of class names provided for in the General Standard is not satisfactory for hypersensitive people. The declaration of allergy-inducing ingredients by general class names may either result in unforeseen hypersensitivity reactions, or in the consumers in question having to refrain from buying food products more often than actually necessary.

3.4. Declaration of carried-over food additives and processing aids

The 25 %-rule clearly states that food additives in the compound ingredient which have a technological function in the finished product shall be declared even though the remaining ingredients in the compound ingredient need not be declared. This is also provided for in Section 4.2.3 of the General Standard, which states:

"4.2.3.1 A food additive carried over into a food in a significant quantity or in an amount sufficient to perform a technological function in that food as a result of the use of raw materials or other ingredients in which the additive was used shall be included in the list of ingredients.

4.2.3.2 A food additive carried over into foods at a level less than that required to achieve a technological function, and processing aids, are exempted from declaration in the list of ingredients.

In other words, food additives with no technological function in the finished product will not appear in the list of ingredients, the same being true for processing aids.

Some food additives may trigger off hypersensitivity reactions at levels below that required to exert a technological function, e.g. sulphites.

Others, e.g. benzoic acid, seem to have a technological function at levels inducing reactions, and shall therefore be declared according to the carry-over principle. However, it is a question as to how this provision is interpreted. The amount of benzoic acid in, for example a fruit preparation used as an ingredient in ice cream may be large enough to exert a preservative effect in theory but not have this effect in practice. Whether or not benzoic acid would be declared in such a case is uncertain.

Concerning other intolerance-inducing food additives like antioxidants, colours, glutamate etc. the relationship between amounts triggering off symptoms and levels considered to have technological effect is unclear.

It is also unclear whether certain processing aids, e.g. nickel and chromium used as catalysts in the hydrogenation of fats, or eggwhite and milk used as clarifying agents in the production of juice, nectar and wine, may give residues which are not tolerated by some consumers. Such substances are not declared on the label, making it difficult to find out whether these residues are of any significance.

Sulphites can induce hypersensitivity reactions at levels of 10 mg/kg, i.e. below the level necessary for technological function.

The significance of other potential intolerance-inducing food additives (e.g. benzoates, azo-colours, antioxidants, glutamate etc.) being present without declaration is less certain and needs further investigation. It is also a question as to how the carry over-principle and the term "technological function" is interpreted and practised.

3.5 Conclusion

In light of the above considerations, it seems quite clear that the General Labelling standard is not satisfactory as regards ensuring that sufficient information is provided to people suffering from food hypersensitivity to enable them to choose products they can tolerate and to avoid products they cannot.

Considering the extent and severity of the problem, it is obviously important that the consumers in question are given the information they need, either by improving labelling requirements or in other ways.

4. POSSIBLE ALTERNATIVES TO THE LABEL DECLARATION OF INGREDIENTS

The essential factor for hypersensitive people is that they are able to make an informed choice. The question could be asked whether the relevant information need always be given on the label, or whether it can be provided in other ways. There are several possibilties.

4.1 Product catalogues, data banks etc.

Product catalogues with detailed information on composition, or lists of products not containing certain ingredients, could be elaborated.

Data banks could be established with detailed product information, with consumers with hypersensitivity in mind. Such data banks already exist to a certain extent with lists of products not containing certain ingredients. Up-dated product catalogues with information from the data bank could be regularly issued.

Product information could also be made available in shops, by means of catalogues, signs, notices etc., or perhaps in the form of leaflets.

Such measures are of great benefit for consumers and not least for dieticians and other health personnel, as an aid in dietary planning.

However, there will always be uncertainty as to whether such information is up-to-date. Product composition may be changed without corresponding updating of the information material. This is likely to be especially a problem with imported products. The wide selection of products on the market further adds to the difficulty of ensuring that information is always correct.

Though data bases are somewhat more reliable in this respect, there is still a considerable risk of erroneous information. As reliance on wrong or incomplete information may have considerable adverse health implications, such systems may in fact aggravate the situation for the consumers concerned, unless it functions properly. This type of information is unlikely to cover the whole range of products, it is unlikely to reach out to all consumers, and it is unlikely that the products included in catalogues are always available to-all consumers.

Data banks, product catalogues etc. are useful supplements to label information, but cannot replace detailed labelling.

4.2 Symbols and logos

The proposal has been made of developing logos or marks to identify products of a certain composition or which are suitable for people with certain kinds of allergy. This idea must be treated with great caution. People suffering from hypersensitivity may react to several food components, the combinations of allergies/intolerances arising varying from person to person. A product which is suitable for one patient may be life-threatening for another.

It would be a difficult, not to say impossible, task to develop symbols which could replace detailed and correct label declarations.

5. IMPROVEMENT OF LABELLING REQUIREMENTS

As stated above, it seems quite clear that the General Labelling Standard is unsatisfactory as regards the declaration of ingredients which can induce hypersensitivity, and that though alternative ways of providing product information may be very useful, they cannot replace detailed labelling. The labelling requirements especially in need of improvement are the 25%-rule and provisions concerning class names. The lacking obligation to declare certain food additives carried over with raw materials and ingredients may also cause some problems.

Incomplete labelling may result in the ingestion of food which may cause considerable discomfort, and at worst serious illness and even death.

The fundamental principle as regards labelling regulations in relation to allergy and intolerance should be that, as far as practically possible, ingredients which are likely to induce hypersensitivity should be declared in the list of ingredients. Special attention must be given to subtances of which small amounts can trigger off severe reactions.

Uncertainty as to whether labelling is correct also results in many people refraining from buying food which they in fact would have tolerated, because they don't know what the product contains, and dare not run the risk of mistakes. The choice of products available to allergic people becomes unnecessarily restricted, and sales of inadequately labelled foods thus reduced.

5.1 The 25%-rule

Ideally, the best solution would be to revoke the 25%-rule, so that section 4. 2. 1. 2 of the General Standard would apply without exception to compound ingredients. However, this might involve practical difficulties. If several compound ingredients are present in small amounts in a product, the ingredient list could well be very long and complex. The declaration would require a lot of space, would be difficult to read, and the information of interest for a person suffering from allergy perhaps difficult to pick out.

Therefore some limitation on the extent to which compound ingredients should be specified is probably necessary. Though this would not be a completely satisfactory for people with hypersensitivity, the increased though still limited degree to which compound ingredients would be specified, would be a step in the right direction. Many more ingredients would be declared, and many more consumers would benefit.

An alternative solution would be to introduce a provision which does not fix a numerical limit, but which lays down guidelines as to when components in a compound ingredient should be declared. For example, a provision stating that ingredients in a compound ingredient shall be declared to the extent necessary to give the consumer a satisfactory picture of the product, and that should a compound ingredient contain components which may represent a potential health problem, e.g. in connection with allergy or intolerance, such components must always be specifically declared.

Such a rule is, however, difficult to implement, as it implies knowledge and awareness on the part of the producer. This is true even if the rule is supplemented by a list of ingredients likely to give rise to health problems. It would also be open to different interpretation in different countries, and thus may give rise to trade barriers.

The most feasible solution would seem to be to reduce the lower limit for specification of compound ingredients as much as is practically possible. Based on an assessment of what is feasible, this lower limit could be set at 5%.

This would trigger off the requirement to specify the components of a compound ingredient in very many cases. However, such a lower limit would still not ensure the declaration of potent allergens which can induce hypersensitive reactions in small amounts. If compound ingredients contain such substances, they should always be declared.

5.2 Class names

As already mentioned, the use of class names may conceal the presence of allergenic ingredients. Class names also create uncertainty for people who for one reason or another must avoid certain kinds of foodstuffs. It would therefore seem best to revoke the possibility to declare ingredients by class name, and require specific declaration in all cases.

On the other hand, however, the use of class names permits a certain degree of freedom to change recipes according to availability of raw materials etc., without having to change the label. Moreover, spice and herb mixtures, for example, in a product are considered by producers to contribute to the special characteristics of the product, and constitute a production secret which they would be very reluctant to divulge. A proposal to revoke the provisions concerning the use of class names will therefore probably meet with vigorous opposition, and may delay the improvements allergic people sorely need. This should rather be a long-term objective. As the next best solution, the withdrawal of some classes should be considered.

Nevertheless, declaration of the most potent allergenic substances must be ensured.

As a minimum solution, a provision should be introduced in the rules governing the use of class names, that states that ingredients which may give rise to health problems in connection with allergy or intolerance may not be declared by class names. A list of ingredients for which specific name declaration is required should be elaborated.

5.3 Declaration of carried-over food additives

The carry-over principle's rules providing for exemption from declaration (i.e. non-functional levels) should not apply to food additives which are known to trigger off hypersensitivity reactions in amounts lower than those giving technological effects. Based on current knowledge, this would only apply to sulphites.

As mentioned in paragraph 4.4, the situation regarding other food additives is less clear. The issue should therefore be examined further.

Food additives which are carried over with raw materials and compound ingredients, and which can induce hypersensitivity reactions in amounts present in the product, should always be declared regardless of whether or not they have technological effect.

5.4 Ingredients which can give rise to hypersensitivity, and which should always be declared no matter the amount present.

As mentioned above, labelling requirements should ensure that ingredients which can induce grave or life-threatening symptoms are always declared, no matter the amounts in which they are present in the product. If all ingredients are not declared by specific name and all components of compound ingredients not specified, special provisions should be elaborated for the declaration of these particular substances. This will, as mentioned, also be necessary even if the the 25%-rule is changed to a 5%-rule.

These ingredients could be declared in association with the class name, or the compound ingredient, as the case may be. Such declarations as "margarine with milk" or "chocolate containing nuts" are obvious examples.

The rules concerning the declaration of compound ingredients and the use of class names must thus include a requirement that the designation of such ingredients shall include the name of the hypersensitivity-inducing substance contained in the ingredient.

It would seem more appropriate to include the name of the component in question in the list of ingredients, rather than as a separate declaration. A special rubric etc. might well give rise to unnecessary apprehension regarding hypersensitivity in consumers not affected by such conditions. The consumers who require special information would know what to look for, and where.

The following ingredients, listed in alphabetic order, can trigger-off reactions in a large number of consumers, and may give rise to serious illness. They should therefore always be declared by specific name in the list of ingredients. The list comprises foods documented to be common causes of allergy and intolerance (7,8,29,30):

Barley, oats, wheat, rye, triticale, and products of these (gluten and starch included)

Crustaceans, shellfish, and products of these

Eggs and egg products

Fish and fish products

Legumes: peas, peanuts, soybeans and products of these

Milk and milk products (lactose included)

Sulphite in concentrations of 10 mg/kg or more

Tree nuts, poppy seeds, sesame seeds and products of these

6. RECOMMENDATIONS

6.1 Amendment of the Codex General Labelling standard

In light of the considerations presented in the document, it is recommended that the Committee (CCFL) discuss the following proposed amendments to the Codex General Labelling Standard (text to be deleted in brackets), text to be added underlined:

Section 4.2.1.3:

Where an ingredient is itself the product of two or more ingredients, such a compound ingredient may be declared, as such, in the list of ingredients, provided that it is immediately accompanied by a list in brackets of its ingredients in descending order of proportion (m/m). Where a compound ingredient (for which a name has been established in a Codex standard or in national legislation) constitutes less than (25%) 5% of the food, the ingredients other than food additives which serve a technological function in the finished product and ingredients known to cause allergic or intolerance reactions need not be declared.

The following foods and ingredients are known to cause hypersensitivity and shall always be declared as such:

Barley, oats, wheat, rye, triticale, and products of these (gluten and starch included)

Crustaceans, shellfish, and products of these

Eggs and egg products

Fish and fish products

Legumes: peas peanuts, soybeans and products of these

Milk and milk products (lactose included)

Sulphite in concentrations of 10 mg/kg or more

Tree nuts, poppy seeds, sesame seeds and products of these

Section 4.2.2.1:

Except for ingredients known to cause reactions of allergy or intolerance such as those listed in Section 4.2.1.3, the following class names may be usedetc., rest of section as is.

Section 4.2.3.2:

A food additive carried over into foods at a level less than that required to achieve a technological function, and processing aids, are exempted from declaration in the list of ingredients. The exemption does not apply to food additives and processing aids known to cause reactions of food allergy or intolerance, such as those listed in in Section 4.2.1.3.

6.2 Further consideration of the issues.

Carried-over food additives and processing aids.

In addition to the above recomendations, it is proposed that the Codex Committee on Food Additives and Contaminants (CCFAC) is requested to assist in assessing labelling requirements for carried-over food additives and processing aids.

CCFAC should be asked to report on what hypersensitivity-inducing food additives and processing aids are likely to be present in a product in amounts which may induce a hypersensitivity response, without, under current requirements, having to be declared. As regards carried-over food additives, the relationship between levels which have a technological function and levels which are required to elicit a reaction will be important. It is also of importance to define precisely how the term "technological function" is to be understood and interpreted in practice and to consider whether declaration of intolerance-inducing food additives should be required above a certain level rather than being tied to the concept of technological function.

Other ingredients

Recommendations concerning other ingredients such as rice, corn(maize), oils, fruits, vegetables, spices, hydrolized proteins etc. require further investigation.

7. LIST OF REFERENCES

- 1) Turner, K.J., Epidemiology of the Allergic Response. CIBA FOUND. SYMP., 1989, 205-29
- 2) Geller-Bernstein C., Levin S., Early-onset Hay Fever in Israeli children. CLIN. ALLERGY 1987, 17(4), 329-32
- Pauli, G. et al, The Assosiation between resipatory allergies and Food Hypersensitivities, ALLERGY AND CLINICAL IMMUNOLOGY NEWS, 1992, 4, 43.
- 4) Veien, N.K., Restriction of Nickel Intake in the treatment of Nickel-sensitive patients, Metabolic Disorders and Nutrition Correlated with Skin, 1991, Vermeer et al (Eds), Basel.
- 5) Kajosaari, M., Food Allergy in Finnish Children Aged 1 to 6 Years ACTA PAEDIATR SCAND 1982; 71:815-819
- Bock, A.S., Prospective Appraisal og Complaints of Adverse Reactions to Foods in Children During the First 3 years of Life, PEDIATRICS, 1987; 79, 5:683-
- 7) Ref in Sampson, H.A., Food Allergy, J. ALLERGY CLIN IMMUNOL 1989;84:1062-1067.
- 8) Fouchard, T. Fodoamnesoverkanslighet, Beskrivningar av allergi och overkanslighet, Sveriges offentlige utredningar 1989:77, 57-68.
- 9) Ref. in Sampson, H.A. Food hypersensitivity: Manifestations, Diagnosis, and Natural History. FOOD TECHN. 1992; :141-144.
- 10) Brostoff, J. og Gamlin, L. Food allergy and Intolerance, THE COMPLETE GUIDE TO FOOD ALLERGY AND INTOLERANCE, 1992, Bloomsbury Publishers, London.
- Young, E. et al, The prevalence of reaction to food additives in a survey population. J R Coll Physicians 1987; 21:241-247
- Fuglesang, G., Madsen, C., Saval, P., Østerballe, O., The Prevalence of Intolerance to Food Additives among Danish School Children. ALLERGY, in press.
- 13) Dahl, R. og Henriksen J.J., Födemiddelallergi og intoleranse, p.42, Symp. Århus 1985
- 14) Simon, R. Allergy proceedings. The XX II CACI Montreux 1988 Schwitzerland
- Bush, R.K. et al, The prevalence of sensitivity to sulfiting agents in asthmatics. AM. J. MED. 1986; 81:816-2
- 15b) Murdoch, D., Pollock, J., Young, E., Lessof, M.H., Food Additive Induced Urticaria: Studies of Mediator Release During Provocation Tests. J R Coll Physicians Lond 1987; 4:262-66
- 16) Cavell, B. et al, Increasing incidence of childhood coeliac disease in Sweden. Results of a national study. ACTA PEDISTR. 1992, 81:589-92
- 17) Rossipal, E., Häufigkeit der Cöliakie im Einzugsgebiet der Kinderklinik Graz. Z. KINDERHEILK, 1975, 119:143-149

- Dahan, S. et al, Coeliac disease in the Rehovot-Ashdod region of Israel: incidence and ethnic distribution. JOURNAL OF EPIDEMIOLOGY AND COMMUNITY HEALTH, 1984, 38:58-60.
- 19) Stanbury, J.B. et al eds, The metabolic basis of inherited disease, Mcgraw-Hill Book Company, chapt. 64.
- 20) Haahtela, T., Hannuksela, M., Food Additives and Hypersensitivity, in FOOD ADDITIVES, Branen, A.L. et al. eds, Marcel Dekker Inc, New York/Basel.
- Smith, M.R. et al, The Role of Food Additives and Intolerance reaction to Food Allergy and Food Intolerance. NUTRITIONAL ASPECTS AND DEVELOPMENTS, Somogy J.C. et al eds., Bibl. Nutri Dieta. Basel, Karger, 1991, 48:72-80.
- Taylor, S.L., Bush, R.K., Nordlee, J.A., Sulfites, in FOOD ALLERGY, Adverse reactions to Foods and Foods Additives: Eds. Metcalfe D.D. et al., Blackwell Scientific Publications, Oxford 1991: 276-287.
- Jacobsen, D.W., Adverse Reactions to Benzoates and Parabens, in FOOD ALLERGY, Adverse reactions to Foods and Foods Additives: Eds. Metcalfe D.D. et al., Blackwell Scientific Publications, Oxford 1991: 276-287.
- Goodman, D.L. et al, Chronic urtocaria exacerbated by the antioxidant food preservatives, butylated hydroxianisole (BHA) and butylated hydroxytoluene (BHT), J. ALLERGY CLIN. IMMUNOL., 1990, 86, 4:570-575
- 25) Stevenson, D.D., Tartratzine, Azo, and Nonazo Dyes, in FOOD ALLERGY, Adverse reactions to Foods and Foods Additives: Eds. Metcalfe D.D. et al., Blackwell Scientific Publications, Oxford 1991: 276-287.
- 26) Allen, D.H., Monosodium Glutamate, in FOOD ALLERGY, Adverse reactions to Foods and Foods Additives: Eds. Metcalfe D.D. et al., Blackwell Scientific Publications, Oxford 1991: 276-287.
- 27) Intolerance to food additives espesially food colours, Codex Committee om food additives, CX/FA 85/5-Add.1.
- Sampson, H.A. and Cooke, S.K., Food Allergy and the Potential Allergenicity-Antigenicity of Microparticulated Egg and Cow's Milk Proteins, JOURNAL OF THE AMERICAN COLLEGE OF NUTRITION, 1990, 9 (4) 410-417.
- Food Additives and Sensitivity, A Scientific Status Summary by the Institute of Food Technologists' Expert Panel on Food Safety & Nutrition, FOOD TECNOLOGY, september 1985, 65-71
- Handlingsplan for barn og unge med allergi/overfølsomhet, astma og andre kroniske lungesykdommer. Helsedirektoratets utredningsserie 2-91. (Directorate of Public Health, Norway 1991)

ALINORM 93/22 APPENDIX V

PROPOSED DRAFT GUIDELINES FOR THE PRODUCTION, PROCESSING, LABELLING AND MARKETING OF ORGANICALLY PRODUCED FOODS (At Step 5 of the Procedure)

Contents

	Page
Foreword	57
1. Scope	59
2. Descriptions and definitions	59
3. Labelling	60
4. Rules of production	62
5. Requirements for inclusion of substances in Annex 2	62
6. Inspection and certification systems	63
7. Imports	65
Annex 1 Principles of organic production - Plants and plant products - Livestock production - Processing, storage and transport	66 66 67 68
Annex 2 Permitted substances for the production of organic foods	70
Annex 3 Minimum inspection requirements and precautionary	76

PROPOSED DRAFT GUIDELINES FOR THE PRODUCTION, PROCESSING, LABELLING AND MARKETING OF ORGANICALLY PRODUCED FOODS

FOREWORD

Background

- 1. Sustainable agriculture represents a broad spectrum of agricultural methodologies which are supportive of the environment. These range from conventional, more intensive methods to alternative methods such as bio-dynamics. Organic agriculture is one method within this range which calls for specific and precise standards of production.
- 2. Organic agriculture is essentially a system based on low external inputs which replace artificial fertilisers and pesticides with an environment that has a high diversity of species and high biological activity. In themselves, organic practices will not ensure that products are completely free of residues of agricultural and veterinary chemicals, and contaminants. It is accepted that pollution from the air, soil, water and other sources is sometimes beyond the control of the operator.
- 3. Requirements for organically produced foods differ from those for other agricultural products in that production procedures are an intrinsic part of the identification and labelling of, and claims for, such products.
- 4. The term "organic" has generally become well understood by those associated with this form of agriculture although in some parts of the world its suitability has been questioned. Other terms have also been introduced such as "biological" and "ecological" in an effort to describe the organic system more clearly. Nevertheless, the term "organic" appears to be the term most widely accepted by the general community.
- 5. For the practical application of organic production methods, more detailed standards are needed to assist the operator in achieving optimal systems which are socially, ecologically and economically sustainable. With the increased interest in organic production, a system of farm evaluation has developed to ensure that products labelled and sold as "organic" actually originate from farms that follow organic production methods. In this way, the consumer is assured of the efficacy of the product and the integrity of the operator is protected.
- 6. Adoption of organic practices requires a period of conversion. This period gives the operator time to adapt to and refine the practices necessary to the environment in which the product is being produced. The system which supports production, ie soil, existing livestock, etc, may also need time for the depletion of possible residues of agricultural chemicals which may exist in the soil, manure heaps, etc and time for livestock to respond to the changed environment.
- 7. The concept of close contact between the consumer and the producer was common in the early days of organic agriculture. It has, however, given way to the introduction of external control and certification procedures stimulated by greater market demand, the increasing economic interests in production, and the increasing distance between producer and consumer.
- 8. An integral component of inspection programs is product certification which provides formal recognition of the operator and contributes to product verification. Procedures for operator certification are based primarily on a yearly description of the agricultural enterprise as prepared by the operator in cooperation with the inspection body. Likewise, at the processing level, standards are also developed against which the processing operations and plant conditions can be inspected and verified. Inspection bodies which certify the procedures of the operator should be independent of economic interests with regard to the certification of operators in order to maintain their integrity.

- 9. Apart from a small portion of agricultural commodities marketed directly from the farm to consumers, most products find their way to consumers via established trade channels. Unfortunately, these are not always free from deceptive practices and transparency of the market is necessary for an audit of the trade and processing enterprises.
- 10. The regulation of a process, rather than a final product, demands responsible action by all involved parties. Generally, it is not possible to fully police the process with inspection staff. Although organic products should be subject to the same testing requirements and standards for safety as conventional products, it is the organic designation which signifies the method of production. To remain credible, the organic industry must be willing to self regulate on an international scale in accordance with internationally adopted guidelines.
- 11. In some countries a number of organic farmer organizations exist. There may be minor differences between their production standards, ideology and regional or personal affiliations although in most cases their aims align very closely. The formation of national "umbrella" organisations enables the whole organic industry to coordinate its activities and to heighten its impact on both the public and the government.
- 12. More recently, some governments have moved to authorise the inspection and certification programs created and operated by inspection bodies. This facilitates government-to-government export certification when required by trading partners and enables competent authorities to verify product.
- 13. These guidelines have been prepared for the purpose of providing an agreed approach to the requirements which underpin production of, and the labelling and claims for, organically produced foods. They take into account the system already introduced in the European Economic Community (EEC), other country developments and the work of the International Federation of Organic Agriculture Movements (IFOAM).
- 14. The aims of these guidelines are:
- to protect consumers against deception and fraud in the market place and unsubstantiated product claims;
- to protect producers of organic produce against misrepresentation of other agricultural produce as being organic;
- to ensure that all stages of production, processing and marketing are subject to inspection and comply with these guidelines;
- to harmonise provisions for the production, certification, identification and labelling of organically grown produce;
- to provide international guidelines for organic food control systems in order to facilitate recognition of national systems as equivalent for the purposes of imports.
- 15. These guidelines set out the principles of organic production at farm, processing, handling, storage and transport stages and provides an indication of accepted permitted inputs for soil fertilising and conditioning, plant and animal pest and disease control and, food additives and processing aids. For labelling purposes, the use of certain terms inferring that organic production methods have been used are restricted to products derived from operators under the supervision of an inspection body.
- 16. Import requirements should be based on the principles of equivalency and transparency as set out in the draft GATT decisions on sanitary and phytosanitary measures. In accepting imports of organic

products, countries would usually assess the inspection and certification procedures and the standards applied in the exporting country.

17. As organic production systems continue to evolve, these guidelines are subject to review and amendment. In particular, the lists of permitted substances for the production of organic foods (Annex 2) are open ones subject to additions and deletions on an ongoing basis. The Codex Committee on Food Labelling (CCFL) will therefore regularly review these lists.

Procedures for revision of these guidelines

18. Member governments and international organisations are invited to make proposals to CCFL regarding amendments and/or additions to these guidelines. Specific provisions for the inclusion of materials in Annex 2 are set out in Section 5 of these guidelines. Proposals should be directed in the first instance to the Chief, Joint FAO/WHO Food Standards Programme, FAO, 00100 Rome, ITALY, (Facsimile No (39) 6 5797 3152) with a copy to the Chairperson, Codex Committee on Food Labelling, Consumer Products Branch, Consumer and Corporate Affairs, 50 Victoria Street, Hull, Quebec KIA OC9, Canada (Facsimile No (1) 819 953 2311).

1. SCOPE

- 1.1 These guidelines apply to the following products which carry, or are intended to carry, descriptive labelling referring to organic production methods:
- (a) unprocessed plants and plant products, animals and unprocessed animal products, and
- (b) processed product for human consumption derived mainly from (a) above.
- 1.2 A product will be regarded as bearing indications referring to organic production methods where, in the labelling, advertising material or commercial documents, the product, or its ingredients, is described by:
 - the terms "organic", "biological", "ecological", "bio-dynamic" or words of similar intent.
- 1.3 Paragraph 1.2 does not apply where these terms clearly have no connection with the method of production.
- 1.4 These guidelines apply without prejudice to other Codex Alimentarius Commission (CAC) provisions governing the production, preparation, marketing, labelling and inspection of the products specified in paragraph 1.1.

2. DESCRIPTION AND DEFINITIONS

2.1 Description

Foods described using the term organic or words of similar intent, are a product of organic farming which is a system of farm design and management practices that seek to create ecosystems which achieve sustainable productivity, and provide weed and pest control through a diverse mix of mutually dependent life forms, recycling plant and animal residues, crop selection and rotation, water management, tillage and cultivation. Soil fertility is maintained and enhanced by a system which optimises soil biological activity as the means to provide a balanced nutrient supply for plant and animal life as well as to conserve soil resources. Pest and disease management is attained by means of the encouragement of a balanced host/predator relationship, augmentation of beneficial insect populations, biological and cultural control and mechanical removal of pests and affected plant parts.

2.2 Definitions

For the purpose of these guidelines:

- (a) "agricultural product" means any product or commodity, raw or processed, that is marketed for human consumption.
- (b) "certified organic farm", or, portion of a farm, or site where agricultural products or livestock are produced, means that such land is certified by a nationally recognised inspection body as utilising a system of organic farming as described in these guidelines;
- (c) "competent authority" means the official government agency having jurisdiction;
- (d) "ingredients" means the substances, including additives, used in the preparation of the products specified in Section 1.1(b) that are still present, albeit in the modified form, in the final product";
- (e) "inspection body" means a body which is responsible for verifying that a product sold or labelled as "organic" is produced, processed, [prepared] and handled according to these guidelines;
- (f) "labelling" means any words, particulars, trademarks, brand names, pictorial matter or symbols, appearing on any packaging, document, notice, label, board or collar accompanying or referring to a product specified in Section 1.1;
- (g) "livestock" means any cattle, sheep, goats, swine, poultry, equine animals used for food or in the production of food; fish used for food; wild or domesticated game, or other non-plant life;
- (h) "marketing" means holding for sale or displaying for sale, offering for sale, selling, delivering or placing on the market in any other form;
- (i) "operator" means any person who produces, prepares or imports, with a view to the subsequent marketing thereof, products as referred to in Section 1.1, or who markets such products;
- (j) "preparation" means cooking, baking, heating, drying, mixing, grinding, churning, separating, extracting, cutting, fermenting eviscerating, preserving, dehydrating, freezing or otherwise manufacturing, and includes packaging, canning, jarring or otherwise enclosing food in a container; or:
 - ["preparation" means the operations of processing, preserving and packaging of agricultural products];
- (k) "production" means the operations involved in producing agricultural products in the state in which they are normally produced on the farm.
- (l) ["veterinary drug" means]

3. LABELLING

- 3.1 Organic products should be labelled in accordance with the Codex General Standard for the Labelling of Prepackaged Foods (CODEX STAN 1-1985).
- 3.2 The labelling and advertising of a product specified in Section 1.1(a) may refer to organic production methods only where:
- (a) such indications show clearly that they relate to a method of agricultural production;

- (b) the product was produced in accordance with the requirements of Section 4 or imported under the requirements laid down in Section 8;
- (c) the product was produced or imported by an operator who is subject to the inspection measures laid down in Section 6.
- 3.3 The labelling and advertising of a product specified in paragraph 1.1(b) may refer to organic production methods only where:
- (a) such indications show clearly that they relate to a method of agricultural production and are linked with the name of the agricultural product in question, as obtained on the farm;
- (b) all the ingredients of agricultural origin of the product are, or are derived from, products obtained in accordance with the requirements of Section 4, or imported under the arrangements laid down in Section 8;
- (c) the product contains only those ingredients of non-agricultural origin as set out in Annex 2, Table 4A;
- (d) the same ingredients shall not be derived from an organic and from a non-organic origin;
- (e) the product or its ingredients have not been subjected during preparation to treatments involving the use of ionizing radiation or substances not listed in Annex 2, Table 4B;
- (f) the product was prepared by an operator subject to the regular inspection system as set out in Section 6 of these guidelines.
- 3.4 By way of derogation from paragraph 3.3(b), certain ingredients of agricultural origin not satisfying the requirement in that paragraph may be used, within the limit of a maximum level of [5%m/m] of the ingredients of agricultural origin in the final product, in the preparation of products as referred to in paragraph 1.1(b)
- providing that such ingredients are of agricultural origin and are not produced in the country or in sufficient quantity in the country in accordance with the requirements of Section 4 of these guidelines.
- 3.5 Information on non-retail containers should be given either on the container or in accompanying documents, except that the name of the product, lot identification, and the name and address of the manufacturer or packer should appear on the container. Lot identification, and the name and address of the manufacturer or packer may be replaced by an identification mark provided that such a mark is clearly identifiable with the accompanying documents.
- 3.6 The labelling and advertising of a product as referred to in paragraph 1.1(b) which has been prepared partly from ingredients not satisfying the production requirements of paragraph 3.3(b) may refer to organic production methods provided that:
- (a) at least [50%] of the ingredients of agricultural origin satisfy the production requirements of paragraph 3.3(b);
- (b) the product satisfies the requirements of paragraphs 3.3(c), (d) and (e);
- (c) the indications referring to organic production methods-

- [appear only in the list of ingredients],
- clearly refer to only those ingredients obtained in accordance with the requirements of Section 4 of these guidelines;
- (d) the ingredients, and the relative levels of the ingredients of agricultural origin, appear in descending order (mass/mass) in the list of ingredients;
- (e) indications in the list of ingredients appear in the same colour and with an identical style and size of lettering.

4. RULES OF PRODUCTION

- 4.1 Organic production methods require that for the production of products referred to in paragraph 1.1(a):
- (a) at least the production requirements of Annex 1 should be satisfied;
- (b) only products composed of substances [or incorporating substances such as those] listed in Annex 2, Tables 1, 2 and 3 may be used as plant protection products, fertilizers, soil conditioners, foliar sprays, animal feedstuffs, or animal protection products insofar as the corresponding use is authorised in general agriculture in the country concerned in accordance with the relevant national provisions.
- 4.2 Organic processing methods require that for the preparation of products referred to in paragraph 1.1(b):
- (a) at least the processing requirements of Annex 1 should be satisfied;
- (b) only products composed of substances or incorporating substances such as those listed in Annex 2, Tables 4A and 4B may be used as ingredients of non-agricultural origin or processing aids insofar as the corresponding use is authorised in the relevant national requirements concerning the preparation of food products and according to good manufacturing practice.

5. REQUIREMENTS FOR INCLUSION OF SUBSTANCES IN ANNEX 2

- 5.1 Countries should develop a list of substances which satisfy the requirements of these guidelines. The following criteria should be used for the purposes of amending these lists of substances not authorized for the purposes indicated in Section 4:
- (a) if they are used for the purpose of plant pest or disease control-
- they are essential for the control of a harmful organism or a particular disease for which other biological, cultural, physical or plant breeding alternatives are not available,
- the conditions for their use [preclude any direct contact with the seed, the crop or crop products; however, in the case of perennial crops, direct contact may take place, but only outside the growing season of the edible parts (fruits) provided that such application] does not indirectly result in the presence of residues of the product in the edible parts,
- their use does not result in, or contribute to, unacceptable effects on, or contamination of, the environment [or have an irreversible influence on local eco-systems];
- (b) if they are used for fertilisation or soil-conditioning purposes-

- they are essential for specific nutrition requirements of crops, or specific soil-conditioning and rotation purposes which cannot be satisfied by the practices mentioned in Annex 1,
- their use does not result in, or contribute to, unacceptable effects on, or contamination of, the environment, [or have an irreversible influence on local eco-systems];
- (c) if they are used for the purpose of animal health or to ensure livestock product quality-
- they are essential for animal health in the advent of a disease outbreak and provided that such animals not be marketed as organic until such time as the residues of the materials have disappeared, provided that other biological, cultural, or physical treatments are not available,
- they do not include growth hormones,
- they are essential for ensuring product quality and preservation and other biological, cultural, or physical treatments are not available;
- (d) if they are used in the production of food-
- they are indispensable for ensuring the safety of the food,
- they are essential to produce or preserve such foods.
- they are preferably nature identical and it is impossible to produce or preserve such food products without having recourse to such ingredients.
- 5.2 Member countries should provide the following for any substance proposed for inclusion in Annex 2:
- (a) a detailed description of the product;
- (b) the conditions of its use and compositional and/or solubility requirements, with regard in particular to the need to insure for these products a minimal presence of residues on edible parts of the crop and on edible crop products or animal products as well as a minimum effect on the environment;
- 5.3 Proposals for amendments to Annex 2, concerning either inclusion or deletion of permitted substances, should be directed in the first instance to the Chief, Joint FAO/WHO Food Standards Programme. Further details for the submission of proposals are provided in the foreword (paragraph 18).

6. INSPECTION AND CERTIFICATION SYSTEMS

- 6.1 Any operator who produces, prepares or imports products as specified in paragraph 1.1 for the purpose of marketing them should:
- (a) give notification of this activity to the competent authority; such notification should include-
 - (i) the name and address of the operator,
 - (ii) the location of the premises and, where appropriate, parcels (land register data) where operations are carried out,

- (iii) the nature of the operation and the products concerned,
- (iv) an undertaking by the operator to carry out the operations in accordance with these guidelines,
- (v) in the case of an agricultural holding, the date on which the producer ceased to apply products not compatible with the production requirements of these guidelines on the parcels concerned,
- (vi) the name of the approved inspection body providing the inspection system;
- (b) submit an undertaking to an inspection system as outlined in this Section.
- 6.2 A competent authority should designate an authority or body for the reception of notifications.
- 6.3 The competent authority should ensure that an updated list containing the names and addresses of operators subject to the inspection system is made available to interested parties.
- 6.4 A country authority should establish an inspection system operated by one or more designated inspection authorities and/or approved private bodies to which the operators producing or preparing products as referred to in paragraph 1.1 should be subject.
- 6.5 A country authority should adopt the measures necessary to ensure that an operator, who complies with the provisions of these guidelines and pays the contribution to inspection expenses, has access to the inspection system.
- 6.6 The inspection system should comprise at least the application of the inspection measures and other precautions set out in Annex 3.
- 6.7 For the application of the inspection system operated by private bodies, countries should designate an authority responsible for the approval and supervision of such bodies. The designated authority may delegate this function to its agent.
- 6.8 For the approval of a private inspection body, the following should be taken into account:
- (a) the standard inspection procedures to be followed, including detailed description of the inspection measures and precautions which the body undertakes to impose on operators subject to its inspection;
- (b) the penalties which the body intends to apply where irregularities are found;
- (c) the availability of appropriate resources in the form of qualified staff, administrative and technical facilities, inspection experience and reliability;
- (d) the objectivity of the body vis-a-vis the operators subject to its inspection.
- 6.9 After an inspection body has been approved, the competent authority should:
- (a) ensure that the inspections carried out by the inspection body are objective;
- (b) verify the effectiveness of its inspections;

- (c) take cognizance of any infringements found and penalties applied;
- (d) withdraw approval of the inspection body where it fails to satisfy the requirements referred to in (a) and (b) or, no longer fulfills the criteria indicated in paragraph 6.6 or, fails to satisfy the requirements laid down in paragraphs 6.10 to 6.12.
- 6.10 The inspection authority and the approved inspection bodies referred to in paragraph 6.4 should:
- (a) ensure that at least the inspection measures and precautions specified in Annex 3 are applied to undertakings subject to their inspection; and
- (b) not disclose information and data obtained in their inspection activities to persons other than the person responsible for the undertaking concerned and the competent public authorities.
- 6.11 Approved inspection bodies should:
- (a) give the competent authority, for [inspection/audit] purposes, access to their offices and facilities, together with any information and assistance deemed necessary by the competent authority for the fulfilment of its obligations pursuant to these guidelines;
- (b) send to the competent authority of the country [by 31 January] each year a list of operators subject to their inspection [on 31 December of the previous year] and present to the said authority a concise annual report.
- 6.12 The inspection authority and inspection bodies referred to in paragraph 6.1 should:
- (a) ensure that, where an irregularity is found in the implementation of Sections 3 and 4, or of the measures referred to in Annex 3, the indications provided for in paragraph 1.2 referring to the organic production method are removed from the entire lot or production run affected by the irregularity concerned;
- (b) where a manifest infringement, or an infringement with prolonged effects, is found, prohibit the operator concerned from marketing products with indications referring to the organic production method for a period to be agreed with the competent authority.

7. IMPORTS

- 7.1 Without prejudice to Section 3, products as specified in paragraph 1.1 which are imported may be marketed only where the competent authority or body in the exporting country has issued a certificate of inspection stating that the lot designated in the certificate was obtained within a system of production and inspection applying rules equivalent to those laid down in these guidelines.
- 7.2 The certificate referred to in paragraph 8.1 above should accompany the goods, in the original copy, to the premises of the first consignee; thereafter the importer should keep the certificate at the disposal of the [inspection authorities/competent authority/inspection body] for not less than two years.
- 7.3 An importing country may:
- (a) require detailed information, including reports established by independent experts, on the measures applied in the exporting country to enable it to make judgements on equivalency;
- (b) conduct on-the-spot examinations of the rules of production and the inspection measures applied in the exporting country.

PRINCIPLES OF ORGANIC PRODUCTION

Plants and plant products

- 1. The principles set out in this Annex should normally have been applied on the parcels during a conversion period of at least two years before sowing, or in the case of perennial crops other than grassland, at least [two/three] years before [the first harvest/the start of the production cycle] of products as referred to in paragraph 1.1(a) of these guidelines. The inspection body may, with the approval of the competent authority, decide, in certain cases, to extend or reduce that period [but not less than 12 months] having regard to previous parcel use.
- 2. [Conversion from conventional to organic production should be effected using permitted techniques as defined in these guidelines, and in accordance with a progressive production plan designed to convert an area of land large enough to permit organic production to be developed and sustained.]
- 3. In cases where a whole farm is not converted at the one time, it may be done progressively, whereby these guidelines are applied from the start of conversion on the relevant fields.
- 4. The fertility and biological activity of the soil should be maintained or increased, where appropriate, by:
- (a) cultivation of legumes, green manures or deep-rooting plants in an appropriate multi-annual rotation programme;
- (b) incorporation in the soil of organic material, composted or not, from holdings producing in accordance with these guidelines. By-products from livestock farming, such as farmyard manure, may be used if they came from livestock holdings producing in accordance with these guidelines;
- (c) appropriate micro-organisms or plant-based preparations (biodynamic preparations) may be used.

Organic or mineral fertilisers, as specified in Annex 2, Table 1 may be applied only to the extent that adequate nutrition of the crop or soil conditioning are not possible by the methods set out in 4(a) and (b) above.

- 5. Pests, diseases and weeds may be controlled by any one, or a combination, of the following measures:
- choice of appropriate species and varieties;
- appropriate rotation programs;
- mechanical cultivation;
- protection of natural enemies of pests through provision of favourable habitat, such as hedges and nesting sites;
- flame weeding;
- biological control [release of predators];

- specific bio-dynamic measures;
- mulching and mowing;
- grazing of livestock;
- [diversified ecosystems. This will vary between geographical locations. For example, in the tropics ecological balancing zones should be established which retain the original vegetations to house pest predators, counteract erosion, etc];
- mechanical controls such as traps, barriers, light and sound;
- [steam sterilization].
- 6. Only in cases of [immediate] threat to the crop may recourse be had to products referred to in Annex 2.
- 7. Seeds and plant propagation material should be from organic production. However, by way of derogation from paragraph 4.1(b), seeds treated with substances not included in Annex 2 but authorised in general agriculture in the country may be used insofar as users of such seed can show to the satisfaction of the inspection body that they were unable to obtain on the market non-treated seed of an appropriate variety of the species in question.

Livestock Production

- 8. Where livestock are maintained, they should be an integral part of the organic farm unit and should be raised and held according to these guidelines.
- 9. Animal products must not be sold as organic unless the animal has been raised according to these guidelines for a period of at least one year.
- 10. Up to 10% of adult animals of a herd or flock may be brought-in annually from non-organic sources for expansion or replacement purposes.
- 11. All brought-in animals from non-organic sources must be produced according to these guidelines for a period of a minimum of one year before their products may be sold under an organic label. Exceptions may be allowed for:
 - (a) calves up to 14 (or 7?) days which have received colostrum and do not come from livestock markets;
 - (b) dairy animals provided that milk is kept separate for a period of 12 (or 4?) weeks;
 - (c) day old poultry; and
 - (d) laying hens, provided that eggs are kept separate for a period of 30 days.
- 12. All livestock systems should be planned to provide the optimum level of 100% of the diet of feedstuffs produced to the requirements of these guidelines;
 - however, by way of derogation, at least [80%/85%] of fodder inputs, calculated on a dry matter basis, should be from organic sources produced in compliance with these guidelines. Exceptions may be granted in cases of extreme climatic or other extenuating circumstances.

- 13. Stocking rates for livestock should be appropriate for the region in question and as regulated by the inspection body for the region.
- 14. [Maintenance of livestock should be guided by an attitude of care, responsibility and respect for living creatures. Pain inflicted by treatments such as castrating, marking and mulesing should be kept to a minimum. Stress should be minimised. Living conditions should consider the natural needs of the animal for free movement, food, water, shelter and shade. Consideration should be given to their specific natural behavioural patterns.]
- 15. [Breeding methods should be in compliance with the principles of organic farming taking into account breeds and strains suitable for raising under local conditions and under an organic system. Own sires should be held. Artificial insemination is not recommended. Embryo transfer techniques are not permitted in the organic farming system.]
- 16. Vaccination of livestock is permitted in cases where a known problem exists or is required by national regulations.
- 17. Vitamins (synthetic), in the absence of natural source vitamins, pure amino acids and trace element supplements are permitted in cases where the need can be demonstrated
- 18. The use of veterinary drugs on livestock in the absence of illness is prohibited. Therapeutic use of veterinary drugs is permitted provided the withholding period is [equal to / double / triple] that required by national legislation for the veterinary drug concerned.
- 19. Growth promotants are prohibited.

[Processing, Storage and Transport]

- 20. The processing of organic food product should meet the requirements of Codex standards and codes of hygienic practice for food production.
- 21. Organic produce may neither be mixed nor substituted with conventional produce.
- 22. Where only part of the unit is certified, other product not covered by these guidelines should be stored and handled separately and both types of products should be clearly identified.
- 23. Products derived from conventional and organic methods should not be stored together, except when packed and handled.
- 24. Bulk stores for organic product should be set aside and clearly labelled to that effect.
- 25. Contamination from any possible non-approved pesticide treatments before using the storage areas shall be excluded.
- 26. Storage areas shall be thoroughly cleaned with methods appropriate to the product.
- 27. Permitted specific storage conditions may include controlled atmosphere (only CO₂, O₂, N₂).
- 28. Pests should be avoided by GMP. Pest control treatment within storage areas may include physical barriers, sound, ultra-sound, light and UV-light; permitted treatments may include:

- traps (including pheromone traps and static bait traps);
- temperature control;
- controlled atmosphere;
- diatomaceous earth.
- 29. All materials used for packaging must conform to food grade packaging materials as established by national regulations.
- 30. In addition, packaging material used for organic products should not contain fungicides, preservatives, or other chemical additives.
- 31. Any food grade packaging material which has previously been in contact with any substance that could compromise the organic quality of the product should not be used.

PERMITTED SUBSTANCES FOR THE PRODUCTION OF ORGANIC FOODS

Precautions

- 1. Any substances used in an organic system for soil fertilisation and conditioning, pest and disease control, for the health of livestock and quality of the animal products, or for processing, preservation and storage of the food product should comply with the relevant national regulations.
- 2. Where substances are required for primary production they should be used with care and with the knowledge that even permitted substances may be subject to misuse and may alter the ecosystem of the soil or farm.
- 3. The lists of ingredients and processing aids of non-agricultural origin included in Table 4 take into account the expectations of consumers that processed products from organic production systems should be composed essentially of ingredients as they occur in nature.

TABLE 1: SUBSTANCES FOR USE IN SOIL FERTILIZING AND CONDITIONING

Substance	Description; compositional requirements; conditions of use
Farmyard and poultry manure	
Slurry or urine	****
Straw	
Peat	
Composts from spent mushroom and	
vermiculture substrates	
Composts from organic household refuse	
Composts from plant residues	
Processed animal products from	
slaughterhouses and fish industries	·
Organic by-products of foodstuffs and textile industries	· .
•••••	
Seaweeds and seaweed products	·
Sawdust, bark and wood waste	
Wood ash	
Natural phosphate rock	
Calcinated aluminium phosphate rock	
Basic slag	
Rock potash	Need recognised by control body
Sulphate of potash	Need recognised by control body
Limestone	
Chalk	
Magnesium rock	
Calcareous magnesium rock	
Epsom salt (magnesium-sulphate)	
Gypsum (calcium sulphate)	 .
Trace elements (boron, copper, iron,	No. 4 was a mind by soutput body
manganese, molybdenum, zinc)	Need recognised by control body
Sulphur	Need recognised by control body
Stone meal	 ,
Clay (bentonite, perlite)	
Homeopathic preparations	
Naturally occurring biological organisms (eg worms)	
Vermiculite	
Peat in seed, potting and module composts only	
Humus from earthworms	

Zeolites

Wood charcoal

TABLE 2: SUBSTANCES FOR PLANT PEST AND DISEASE CONTROL

Substance

Description; compositional requirements; conditions of use

Preparations on basis of pyrethrins extracted from

Chrysanthemum cinerariaefolium, containing possibly a synergist

Preparations from Derris elliptica

Preparations from Quassia amara

Preparations from Ryania speciosa

Propolis

Diatomaceous earth

Stone meal

Preparations on basis of metaldehyde containing a repellent to higher

animal species and as far as applied within traps

Sulphur

Bordeaux mixture

Burgundy mixture

Sodium silicate

Sodium bicarbonate

Potassium soap (soft soap)

Pheromone preparations

Bacillus thuringiensis preparations

Granulose virus preparations

Plant and animal oils

Paraffin oil

Seaweed, seaweed meal, seaweed extracts,

sea salts and salty water

Homeopathic preparations

Neem oil and extracts

Natural plant extracts, excluding tobacco

Potassium permanganate

Carbon dioxide and nitrogen gas

Vinegar

Mineral powders

Herbal and bio-dynamic preparations

TABLE 3: SUBSTANCES FOR ANIMAL PEST AND DISEASE CONTROL

Substance	Description; compositional requirements; conditions for use
Pyrethrum extracted from Chrysanthemum cinerariaefolium,	
Rotenone extracted from Derris elliptica	
Quassia extracted from Quassia amara	
Neem oil and extracts	
Garlic oil, garlic extract or crushed garlic	
Seaweed, seaweed meal, seaweed extracts,	
sea salts and salty water	
Sulphur	
Potassium permanganate	
Homeopathic preparations	
Natural plant extracts obtained by infusion, excluding tobacco	
Essential oils	
Methylated spirits	
Tallow	
Cidar vinegar (certified organic)	·
Nettle	
Diatomaceous earth (non heat-treated form)	
Selenium and other trace elements	
Zinc sulphate	

Copper sulphate

Vaccines

TABLE 4A: INGREDIENTS OF NON-AGRICULTURAL ORIGIN REFERRED TO IN SECTION 3 OF THESE GUIDELINES

A1. Food additives, including carriers

INS	Name	Specific conditions
170	Calcium carbonates	specific conditions
270	Lactic acid	
290	Carbon dioxide	
296	Malic acid	
300	Ascorbic acid	
322	Lecithin	
330	Citric acid	
334	Tartaric acid	
335	Sodium tartrate	
336	Potassium tartrate	
400	Alginic acid	
401	Sodium alginate	
402	Potassium alginate	
406	Agar	
410	Locust bean gum	
412	Guar gum	
413	Tragacanth gum	
414	Arabic gum	
416	Karaga gum	
440	Pectins (unmodified)	
500	Sodium carbonates	
501	Potassium carbonates	7000
503	Ammonium carbonates	
504	Magnesium carbonates	
516	Calcium sulphate	Carrier
938	Argon	
941	Nitrogen	
948	Oxygen	****
	• •	

A2. Flavourings

Substances and products labelled as natural flavouring substances or natural flavouring preparations as defined in CAC/Vol XIV - Ed 1. Supplement 1.

A3. Water and salts

- Drinking water
- Salts (with sodium chloride or potassium chloride as basic components generally used in food processing).

A4. Preparations of Microorganisms

- (a) Any preparations of microorganisms normally used in food processing, with the exception of microorganisms genetically modified;
- (b) Microorganisms genetically modified if they have been included according to the decision procedure of Section 5 of these guidelines.

A5. Minerals (including trace elements) and vitamins

Only approved in so far as their use is legally required in the food products in which they are incorporated.

TABLE 4B: PROCESSING AIDS WHICH MAY BE USED FOR THE [PROCESSING/PREPARATION] OF PRODUCTS OF AGRICULTURAL ORIGIN REFERRED TO IN SECTION 3 OF THESE GUIDELINES

Name	Specific conditions	
Water		
Calcium chloride	coagulation agent	
Calcium carbonate	••••	
Calcium hydroxide	••••	
Calcium sulphate	coagulation agent	
Magnesium chloride (or nigari)	coagulation agent	
Potassium carbonate	drying of raisins	
Carbon dioxide		
Nitrogen		
Ethanol	solvent	
Tannic acid	filtration aid	
Egg white albumin		
Casein	· ••••	
Gelatin		
Isinglass		
Vegetable oils	greasing or releasing agent	
Silicon dioxide (gel) or colloidal solution	xide (gel) or colloidal solution	
Activated carbon		
Talc		
Bentonite		
Kaolin		
Diatomaceous earth		
Perlite		
Hazelnut shells		
Beeswax	releasing agent	
Carnauba wax	releasing agent	

Preparations of microorganisms and enzymes:

- (a) Any preparations of microorganisms and enzymes normally used as processing aids in food processing, with the exception of genetically modified organisms and enzymes;
- (b) Microorganisms genetically modified if they have been included hereunder according to the decision procedure of Section 5 of these guidelines.

MINIMUM INSPECTION REQUIREMENTS AND PRECAUTIONARY MEASURES UNDER THE INSPECTION SYSTEM

A. Production at farm level

- 1. Production should take place in a unit where the land parcels, production areas and storage facilities are clearly separate from those of any other unit which does not produce according to these guidelines; processing and/or packaging workshops may form part of the unit, where its activity is limited to processing and packaging of its own agricultural produce.
- 2. When the inspection arrangements are first implemented, the operator and inspection body should draw up:
- a full description of the unit, showing the storage and production premises and land parcels and, where applicable, premises where certain processing and/or packaging operations take place;
- all the practical measures to be taken at the level of the unit to ensure compliance with these guidelines.

This description and the measures concerned should be contained in an inspection report countersigned by the responsible person of the unit. In addition, the report should specify:

- the date of the last application on the land parcels concerned of products the use of which is not compatible with Section 4 of these guidelines;
- an undertaking by the operator to carry out operations in accordance with Sections 3, 4 and 8 and to accept, in event of infringements, implementation of the measures as referred to in paragraph 6.12 of these guidelines.
- 3. Each year, before the date indicated by the inspection body, the operator should notify the body of its schedule of production of crop products [and livestock], giving a breakdown by land parcel [/herd].
- 4. Written and/or documentary accounts should be kept which enable the inspection body to trace the origin, nature and quantities of all raw materials bought, and the use of such materials; in addition, written and/or documentary accounts should be kept of the nature, quantities and consignees of all agricultural products sold. Quantities sold directly to the final consumer should be accounted for on a daily basis.
- 5. Storage, on the unit, of input substances, other than those whose use is compatible with paragraph 4.1(b) of these guidelines is prohibited.
- 6. Apart from unannounced inspection visits, the inspection body should make a full physical inspection, at least once a year, of the unit. Samples for testing of products not authorised [listed]

in these guidelines may be taken. Such samples should be taken where the use of unauthorised products is suspected. An inspection report should be drawn up after each visit and countersigned by the person responsible for the unit.

- 7. The operator should give the inspection body, for inspection purposes, access to the storage and production premises and to the parcels of land, as well as to the accounts and relevant supporting documents. The operator should also provide the inspection body with any information deemed necessary for the purposes of the inspection.
- 8. Products referred to in Section 1 of these guidelines which are not in their packaging for the end consumer should be transported in a manner which would prevent substitution of the content and provided with a label stating, without prejudice to any other indications required by law:
- the name and address of the person responsible for the production or preparation of the product;
- the name of the product;
- that the product is covered by an inspection arrangement equivalent to those set out in these guidelines.
- 9. Where an operator runs several production units in the same area, units in the area producing crop, crop products [or livestock] not covered by Section 1 should also be subject to the inspection arrangements as regards the dash points of paragraph 2 and paragraphs 3 and 4 above. Plants [and animals] of the same type as those produced at the unit referred to in paragraph 1 above should not be produced at these units.

B. Processing and packaging units

- 1. When the inspection arrangements are first implemented, the producer and [inspection body] should draw up:
- a full description of the unit, showing the facilities used for the processing, packaging and storage of agricultural products before and after the operations concerning them;
- all the practical measures to be taken at the level of the unit to ensure compliance with these guidelines.

This description and the measures concerned should be contained in an inspection report, countersigned by the responsible person of the unit.

In addition, the report should include an undertaking by the operator to perform the operations in such a way as to comply with Section 4 of these guidelines and to accept, in the event of infringements, the implementation of measures as referred to in paragraph 6.12 of these guidelines.

2. Written accounts should be kept enabling the inspection body to trace:

- the origin, nature and quantities of agricultural products as referred to in Section 1 of these guidelines which have been delivered to the unit;
- the nature, quantities and consignees of products as referred to in Section 1 of these guidelines which have left the unit;
- any other information such as the origin, nature and quantities of ingredients, additives and manufacturing aids delivered to the unit and the composition of processed products, that is required by the inspection body for the purposes of proper inspection of the operations.
- 3. Where products not referred to in Section 1 of these guidelines are also processed, packaged or stored in the unit concerned:
- the unit should have separate areas within the premises for the storage of products as referred to in Section 1 of these guidelines, before and after the operations;
- operations should be carried out continuously until the complete run has been dealt with, separated by place or time from similar operations performed on products not covered by Section 1 of these guidelines;
- if such operations are not carried out frequently, they should be announced in advance, with a deadline agreed on with the inspection body;
- every measure should be taken to ensure identification of lots and to avoid mixtures with products not obtained in accordance with the requirements of these guidelines.
- 4. Apart from unannounced inspection visits, the inspection body should make a full physical inspection, at least once a year, of the unit. Samples for testing of products not authorized under these guidelines may be taken. However, they should be taken where the use of unauthorized products is suspected. An inspection report must be drawn up after each visit countersigned by the person responsible for the unit inspected.
- 5. The operator should give the [inspection body], for inspection purposes, access to the unit and to written accounts and relevant supporting documents. The operator should also provide the [inspection body] with any information necessary for the purposes of inspection.
- 6. The requirements in respect to the transport as laid down in paragraph A.8 of this Annex are applicable.